POLICY RESEARCH WORKING PAPER

2383

New Firm Formation and Industry Growth

Does Having a Market- or Bank-Based System Matter?

Thorsten Beck Ross Levine Do industries that depend heavily on external finance grow faster in market-based or bank-based financial systems? Are new firms more likely to form in a bank-based or a market-based financial system?

The World Bank Financial Sector Strategy and Policy Department June 2000



POLICY RESEARCH WORKING PAPER 2383

Summary findings

Beck and Levine find no evidence for the superiority of either market-based or bank-based financial systems for industries dependent on external financing. But they find overwhelming evidence that industries heavily dependent on external finance grow faster in economies with higher levels of financial development and with better legal protection for outside investors — including strong creditor and shareholder rights and strong contract enforcement mechanisms.

Financial development also stimulates the establishment of new firms, which is consistent with the Schumpeterian view of creative destruction.

Financial development matters. That the financial system is bank-based or market-based offers little additional information.

This paper — a product of the Financial Sector Strategy and Policy Department — is part of a larger effort in the department to understand the link between financial development and economic growth. Copies of the paper are available free from the World Bank, 1818 H Street NW, Washington, DC 20433. Please contact Elena Mekhova, room MC9-622, telephone 202-458-5984, fax 202-522-2031, email address emekhova@worldbank.org. Policy Research Working Papers are also posted on the Web at www.worldbank.org/research/workingpapers. The authors may be contacted at tbeck@worldbank.org or rlevine@csom.umn.edu. June 2000. (43 pages)

The Policy Research Working Paper Series disseminates the findings of work in progress to encourage the exchange of ideas about development issues. An objective of the series is to get the findings out quickly, even if the presentations are less than fully polished. The papers carry the names of the authors and should be cited accordingly. The findings, interpretations, and conclusions expressed in this paper are entirely those of the authors. They do not necessarily represent the view of the World Bank, its Executive Directors, or the countries they represent.

New Firm Formation and Industry Growth: Does Having a Market- or Bank-Based System Matter?

Thorsten Beck and Ross Levine

JEL Classification: G1; G2; O4

Keywords: Financial Structure; Economic Growth; External Finance

Beck: World Bank; Levine: Carlson School of Management, University of Minnesota. This paper's findings, interpretations, and conclusions are entirely those of the authors and do not necessarily represent the views of the World Bank, its Executive Directors, or the countries they represent. Work for this paper was completed while both authors were visiting the Banco Central de Chile, which provided an excellent research environment. We would like to thank Franklin Allen, Nicola Cetorelli, Robert Cull, Patrick Honohan, Asli Demirgüç-Kunt, Raghuram Rajan, and Rene Stulz for helpful comments.

•		
· .		

1. Introduction

An intensive debate focuses on the relative merits of bank-based versus market-based financial systems. Many authors stress the advantages that banks have over markets in financing the expansion of existing firms and in promoting the establishment of new firms. Others, however, emphasize the comparative merits of markets. Historically, empirical research on the relative merits of bank-based versus market-based financial systems has centered on Germany and Japan as bank-based systems and the United States and Great Britain as market-based financial systems. This work has produced illuminating insights concerning the operation of financial systems in these countries. Nevertheless, it is very difficult to draw broad conclusions about bank-based and market-based financial systems from only four countries. To ameliorate this shortcoming, we have compiled a broad cross-country dataset with measures of the degree to which countries have bank-based or market-based financial systems.

Using this new data, this is the first paper to examine the following questions:

- (1) Do industries that depend heavily on external finance grow faster in bank-based or market-based financial systems?
- (2) Are new firms more likely to form in a bank-based or market-based financial system? Thus, this paper concentrates on specific mechanisms through which financial structure influences economic activity and industrial structure.

This paper empirically assesses the validity of four competing theoretical perspectives regarding bank-based and market-based financial systems in the process of economic development. The *bank-based view* highlights the positive role of banks in providing external

finance and funding new firms. First, a large literature explores the positive role of banks without suggesting that banks have a comparative advantage over markets in providing external finance. Banks may reduce the costs of screening and monitoring firms and managers and thereby improve resource allocation and corporate control [Diamond 1984; Ramakrishnan and Thakor 1984; Boyd and Prescott 1986]. Furthermore, banks frequently establish close, long-term relations with firms and ease cash-flow constraints on existing firm expansion with positive ramifications on economic growth [Hoshi, Kashyap, and Scharfstein 1991]. Similarly, banks are effective in providing external resources to new firms that require staged financing. By forming long-run relationships, banks can credibly commit to making additional funding available as the project develops [Stulz, 2000]. Thus, in arising to mitigate information and transaction costs, banks may promote the expansion of existing firms and the creation of new ones.

The bank-based view, however, not only argues that banks are important for easing the flow of external finance. The bank-based view also critiques the role of markets and emphasizes the comparative merits of banks. First, Stiglitz (1985) argues that well-developed markets quickly reveal new information to all investors. This reduces the incentives for individual investors to expend resources researching firms because any new information they uncover is quickly reflected in stock prices. Thus, market-based systems may hinder incentives for acquiring information. Bank-based systems mitigate this problem since banks establish long-run commitments to firms and reveal less information in public markets [Boot, Greenbaum, and Thakor 1993]. Furthermore, greater market liquidity makes it easier for shareholders to sell their shares and cut their ties to firms. This reduces the incentives for shareholders to expend

¹ See Allen and Gale (1999), Boot and Thakor (1997), Gerschenkron (1962), Goldsmith (1969), Levine (2000b), Rajan and Zingales (1999), Stiglitz (1985), and Stulz (2000) for discussion and additional references regarding the relative merits of bank- and market-based financial systems in fostering economic performance.

substantial resources exerting corporate control [Shleifer and Vishny, 1986].² Thus, greater market development might hinder external financing and economic growth by reducing the effectiveness of corporate control. Also, incestuous relationships frequently flourish between management and boards of directors, which may induce directors and management to collude again other shareholders [Allen and Gale, 1999]. Thus, market-based systems may be less effective than bank-based systems in assessing information about firms and managers and therefore at providing external finance to firms.

The *market-based view* stresses the advantages of markets in funding firms and promoting new, innovative enterprises. Again, there are two parts to the market-based view: the first part emphasizes the positive role of markets and the second highlights the comparative advantages of markets relative to banks. A big and liquid market aggregates views on new technologies and displays this in public prices. Furthermore, in liquid markets, agents that obtain valuable information can quickly profit by trading in these markets, which in turn stimulates market participants to acquire information about firms.³ Stock markets may also stimulate greater corporate control by facilitating takeovers and making it easier to tie managers' compensation to firm performance. Thus, market-based proponents argue that well-functioning markets stimulate information acquisition and corporate control that in turn spur the growth of new and existing firms.

Proponents of the market-based view also emphasize the advantages of market-based systems relative to bank-based ones. First, by acquiring inside information about firms, banks

² Atomistic shareholders also have incentives to capture the benefits from a takeover by holding on to their shares instead of tendering them, thus making takeover attempts less profitable and therefore a less useful control device [Grossman and Hart, 1980]. Also, corporate control through outside takeover threats may face similar limitations because insiders have greater information than outsiders.

³ Allen and Gale (1999) show that riskier, more innovative industries will tend to find it easier to attract external resources in market-based financial systems. These industries might be the more innovative ones. Allen (1993)

can extract informational rents from firms [Hellwig, 1991]. The banks' market power reduces the incentives of firms to undertake profitable projects since banks extract a large share of the profits [Rajan, 1992]. ⁴ By encouraging competition, market-based systems create greater incentives for entrepreneurship than bank-based systems. Second, bankers tend to be ineffective corporate controllers due to their insider status. In bank-based systems, bankers often hold equity and vote the shares of other shareholders. Thus, bankers might collude with managers against other outside investors and thereby thwart competition and hinder effective corporate control [Hellwig, 1998; Wenger and Kaserer, 1998]. Third, banks – as debt issuers – have an inherent bias toward conservative investments, so that bank-based systems stymie innovation and growth. Weinstein and Y afeh (1998) and Morck and Nakamura (1999) find evidence of this in Japan. Also, in the absence of sound, market price signals, banks might continue financing established firms with low-return projects [Rajan and Zingales, 1999]. Thus, relative to market-based systems, bank-based systems may be less likely to funnel external finance to the newest, most profitable endeavors.

The *financial services view* argues that the bank-based versus market-based debate is of second-order importance. According to this view, both banks and markets arise to ameliorate information and transactions costs and thereby provide financial services. In comparing countries, the first-order issue is the quality and availability of financial services, not whether banks or markets provide these services. Furthermore, banks and markets might act as complements in providing financial services and promoting economic development [Boyd and

argues that market-based economies such as the U.S. have been better in developing new and innovative industries than bank-based economies, such as Germany or Japan.

⁴ Petersen and Rajan (1995), however, present evidence that small businesses in the U.S. are less credit-constrained in more concentrated banking markets. Cetorelli and Gamberra (2000) find that financially more dependent industries grow faster in economies with more concentrated banking sectors.

Smith 1998; Huybens and Smith 1999].⁵ According to the financial services view, it is not bank-based or market-based systems per se, but rather the overall level of financial development that stimulates the flow of external funds to worthy firms.

The *legal-based view* [LaPorta, Lopez-de-Silanes, Shleifer, and Vishny 1999b] rejects the analytical relevance of the bank-based versus market-based debate altogether. This view instead argues that countries with legal codes that protect outside investors and legal systems that enforce those codes will have financial systems that facilitate external finance. Thus, the legal-based view predicts that the component of overall financial development defined by the legal system critically influences the expansion of existing firms and the formation of new ones.

To evaluate the competing predictions of the bank-based, market-based, financial services, and legal-based views, we use a panel of 42 countries and 36 industries over the 1980s. As in Rajan and Zingales (1998) and Cetorelli and Gamberra (2000), we focus on industries and decompose industrial growth into that part due to the creation of new firms and that part due to the expansion of existing firms. Rajan and Zingales (1998, henceforth RZ) show that industries that are externally dependent - industries that are naturally heavy users of external finance – grow relatively faster in economies with higher levels of financial development. Thus, as a first step, we examine whether externally dependent industries grow faster in market- or bank-based financial systems, or whether it is the overall level of financial development that enhances the growth of externally dependent industries. We also assess whether the legal system plays a critical role in producing a financial system that fosters the growth of externally dependent firms. In a second step, we decompose industry growth into the growth of the number of establishments

⁵ Also, see Stulz (2000) who notes that stock markets offer an exit option for entrepreneurs and therefore decreases banks' market power. Stock markets also enable an entrepreneur who has obtained bank financing to realize profits from a successful project by selling it. The existence of markets increases the return for the entrepreneurs and thus the incentives to undertake risky, innovative projects. Together, banks and markets work to promote growth.

and the growth of the average size of establishments. Thus, we assess whether bank-based or market-based financial systems are better at fostering new firm formation or existing firm expansion, or whether it is the overall level of financial development that is critical for the emergence of new firms and the expansion of existing ones. Moreover, we also examine whether the legal system plays the pivotal role in establishing the financial conditions for both new firm formation and the expansion of old firms.

The results give no support to either the market- or bank-based view. Industries that depend heavily on external finance do not grow faster in either bank-based or market-based financial systems. The results are supportive of the financial-services and the legal-based views. Industries that depend heavily on external finance grow faster in economies with higher levels of overall financial development. Industries that depend heavily on external finance also grow comparatively faster in economies where legal codes protect the rights of outside investors and where the legal system effectively enforces those codes. We then decompose the industry growth rates into the growth in the number of establishments and the growth in the average size of establishments. We again find that there is not a robust relationship between the degree to which a country has a bank-based or market-based financial system and the rate of new firm formation or existing firm expansion. In contrast, overall financial development explains crosscountry variation in the growth in the number of establishments. Similarly, the legal environment governing financial contracting helps explain cross-country variation in the growth in the number of establishments. Neither overall financial development nor the legal environment, however, is significantly associated with the growth in the average size of establishments. Thus, overall financial development and the legal environment are critically

important for new firm creation, which is consistent with Schumpeter's view of the role of the financial system in fostering creative destruction [King and Levine 1993].

This paper is importantly different from two recent papers on financial structure and economic growth. Levine (2000b) shows that financial structure is not a good predictor of growth in a cross-country growth framework: neither bank-based nor market-based financial systems are closely associated with economic growth. He, however, examines Gross Domestic Product (GDP) growth. He does not examine whether financial structure influences new firm creation, existing firm expansion, or external financing, which is this paper's focus.

Furthermore, Demirgüç-Kunt and Maksimovic (2000) use firm-level data and also show that financial structure is not a robust predictor of economic growth. Again, however, they do not examine whether financial structure influences new firm creation or existing firm expansion, which are two channels highlighted by the theoretical literature discussed above.

The remainder of this paper is organized as follows. Section 2 describes the econometric model that we use to evaluate the comparative ability of the (1) bank-based, (2) market-based, (3) financial services, and (4) legal-based views to explain industrial expansion and new firm creation. Section 3 presents the data. Section 4 provides and discusses the empirical results. Section 5 presents sensitivity analyses and section 6 concludes.

2. Methodology

Financial intermediaries and markets help overcome market frictions that drive a wedge between the price of external and internal finance. Lower costs of external finance facilitate firm growth and new firm formation. Therefore, industries that are naturally heavy users of external finance should benefit disproportionately more from greater financial development than industries that are not naturally heavy users of external finance.

RZ (1998) find evidence consistent with the hypothesis that industries that rely more heavily on external finance grow faster in countries with a better-developed financial system. They use data on a panel of 42 countries and 36 industries over the 1980s. Furthermore, RZ show that the effect of financial development on the industrial growth runs mostly through growth in the number of establishments rather than through growth in the average size of establishments. So financial development improves disproportionately the prospects of young firms in industries that rely heavily on external finance.

Besides confirming the RZ findings with alternative measures of financial development and alternative instruments to deal with endogeneity, this paper explores whether industries with a high need of external finance grow faster in economies with bank- or market-based financial systems. Furthermore, we assess the financial services and legal-based views. Specifically, we study whether (1) the overall level of financial development and (2) the legal codes and enforcement mechanisms governing financial transactions importantly determine industrial growth patterns. Following RZ, we then decompose industry growth into the growth in the average size of establishments and the growth in the numbers of establishments to evaluate the importance of the degree to which financial systems are bank-based or market-based, financial development and its legal determinants for the sources of growth. For conciseness, we use the term "financial structure" to refer to the degree to which a country's financial system is bank-based or market based. We define our indicators of financial structure in such a way that higher values imply a more market-based system.

Econometrically, we use the following regression to assess the impact of financial development and financial structure on industry growth.

 $6 \text{ rowth}_{i,k} = \alpha_j Country_j + \beta_l Industry_l + \gamma Share_{i,k} + \delta_l (External_k * FD_i) + \delta_2 (External_k * FS_i) + \varepsilon_{l,i}$ where $6 \text{ rowth}_{i,k}$ is the average annual growth rate of value added, the growth in number of firms or the growth in average size of firms, in industry k and country i. Country and Industry are country and industry dummies, respectively, and $Share_{i,k}$ is the share of industry k in manufacturing in country i in 1980. External_k is the measure of dependence on external finance for industry k as measured for a sample of U.S. companies over the period 1980-89. FD_i and FS_i are indicators of financial development and financial structure for country i, respectively. We interact the external dependence of an industry (External) with both (a) a measure of overall financial development (FD) and (b) an index of the degree of market-based versus bank-based, i.e., an index of financial structure (FS). The dummy variables for industries and countries correct for country and industry specific characteristics that might determine industry growth patterns. We thus isolate the effect that the interaction of external dependence and financial development/structure has on industry growth rates relative to country and industry means. By including the initial share of an industry we control for a convergence effect; we expect industries with a large share to grow more slowly, and therefore a negative sign on γ .

The different hypotheses imply different predictions about the sign and significance of δ_1 and δ_2 . The *market-based view* predicts that industries that are dependent on external finance grow faster in economies with market-oriented financial systems and higher levels of financial development, thus implying $\delta_1 > 0$ and $\delta_2 > 0$. The *bank-based view* predicts that industries that are dependent on external finance grow faster in economies with bank-oriented financial systems

⁶ We do not include Financial Development or Financial Structure on their own, since we focus on within-country, across-industry growth rates.

⁷ This does not correspond exactly to the convergence concept known from cross-country growth regressions. We include the share in manufacturing rather than the level, since we focus on within-country, across-industry growth rates. As in RZ, γ enters significantly negative in most regressions.

and higher levels of financial development, thus implying δ_1 > 0 and δ_2 < 0. The *financial-services* view predicts that industries dependent on external finance grow faster in economies with a higher level of overall financial development, whereas the financial structure should not matter, thus implying δ_1 > 0 and δ_2 =0. The *legal-based view* predicts that industries dependent on external finance grow faster in economies that protect the rights of outside investors more efficiently, whereas financial structure should not matter. If we replace FD_i with indicators of these legal rights and contract enforcement, this implies δ_1 > 0 and δ_2 =0.

We run both Ordinary-Least-Squares (OLS) regressions and Two-Stage-Least-Squares (TSLS) regressions. TSLS regressions allow us to address the issue of endogeneity of independent variables. Specifically, we want to control for the possible endogeneity of the level and the structure of financial development. Whereas the above equation suggests that an exogenously given level or structure of financial sector activity might interact with the external dependence of industries to determine industry growth rates, financial markets and institutions might have arisen due to a given industrial structure.

By using appropriate instruments we can control for simultaneity bias and reverse causality. We will use the legal origin and the religious composition of countries as instrumental variables for the level and structure of financial sector development. Legal systems with European origin can be classified into four major legal families [Reynolds and Flores, 1996]: the English common law and the French, German, and Scandinavian civil law countries. Most countries have acquired their legal systems through occupation and colonialism, so that the legal origin can be regarded as exogenous. Furthermore, LaPorta, Lopez-de-Silanes, Shleifer and Vishny (1997, 1998) have shown that the legal origin of a country materially influences its legal treatment of creditors and shareholders, its accounting standards and the efficiency of contract

enforcement. Since these regulatory and informational characteristics determine the efficiency of financial intermediaries and markets, we regard the legal origin of countries as good instruments for financial development. LaPorta, Lopez-de-Silanes, Shleifer, and Vishny (1999a) also show that the dominant religion of a country influences institutional development.

3. The Data

This section describes the measure of external dependence, the indicators of financial development and structure and the industrial growth data. The data are for 42 countries and 36 industries. All industries are in manufacturing as in RZ.

3.1. External Dependence

The industry-level data on external dependence are from RZ (1998). The underlying assumption in RZ – and our work — is that for technological reasons some industries depend more heavily on external finance than others. Scale economies, gestation period or intermediate product intensity might constitute some of these technological reasons. Unfortunately, we can only observe the actual use of external finance, but not the demand for it. If financial markets were relatively frictionless, the actual use of external finance would represent the equilibrium of supply and demand. For countries with very well developed financial systems, RZ note that external funds will be supplied very elastically, so that the actual use of external finance would primarily reflect the demand for external finance. Assuming that the variance of the need of external finance across industries persists across countries we can thus use the actual external dependence of industries as observed in a country with a very well developed financial system as a proxy for the "natural" dependence of industries on external finance. As discussed in RZ, we

use the United States to compute the natural external dependence and then we confirm our results using Canadian data to compute the natural external dependence of industries.

The data are from Standard and Poor's *Compustat* for U.S. firms in 36 industries. This database contains only publicly listed firms. A firm's dependence on external finance is defined as the share of investment that cannot be financed through internal cash flows; or as capital expenditures minus cash flow from operations divided by capital expenditures. Both numerator and denominator are averaged over the 1980s to smooth temporal fluctuations. The industry values are calculated as medians rather than means to thus prevent outliers from dominating the results. Table 1 lists the external dependence for all 36 industries. The drug industry is the industry most dependent on external finance, whereas the tobacco industry has no demand for external finance, i.e. our dependence measure is less than zero. Table 1 also lists three alternative measures of external dependence that we will discuss in the section on robustness tests.

3.2. Indicators of Financial Development and Structure

3.2.1 Indicators of Financial Development

To test our hypotheses, we need appropriate indicators of the efficiency with which financial intermediaries and markets reveal information and exert corporate control, and therefore channel external resources to industries that need them most. While the perfect indicators certainly do not exist, the recent literature has developed indicators that proxy relatively well for financial intermediary and stock market development across countries. We use newly constructed data in Beck, Demirgüç -Kunt, and Levine (2000) and Levine, Loayza, and Beck (2000) to measure overall financial development. We also construct new measures of financial development, isolating private financial intermediaries, and discuss these results in the sensitivity section below.

Finance-Activity is a measure of the overall activity of the financial intermediaries and markets. It is defined as the log of the product of Private Credit, the value of credits by financial intermediaries to the private sector divided by GDP, and Value Traded, the value of total shares traded on the stock market exchange divided by GDP. Private Credit is the most comprehensive indicator of the activity of financial intermediaries by including both bank and nonbank intermediaries. Recent work shows that Private Credit exerts a large, positive, robust influence on economic growth [Levine, Loayza, and Beck 2000; and Beck, Levine, and Loayza 2000]. Value Traded measures the activity of the stock market trading volume as a share of national output and thus indicates the degree of liquidity that stock markets provide to economic agents. 8

⁸ Levine and Zervos (1998) point out a potential pitfall of Value Traded. If forward-looking stock markets anticipate large corporate profits and therefore higher economic growth, this will boost stock prices and therefore boost Value Traded. Thus, a positive relationship between Value Traded and growth might reflect a spurious correlation due to this price effect. This price effect, however, does not arise in our model, since we focus on within-country, across-industry growth rates. If markets anticipate higher growth in one industry, the resulting larger value of Value Traded

Finance-Size is a measure of the overall size of the financial sector and is defined as the log of the sum of Private Credit and Market Capitalization. Market Capitalization is defined as the value of listed shares divided by GDP, and is a measure of the size of stock markets relative to the economy. While we include this in our analysis, past work suggests that market capitalization is not a very good predictor of economic performance (Levine and Zervos 1998).

Finance-Aggregate combines the previous two measures and is thus a conglomerate indicator of the size and activity of the financial sector. Specifically, it is the first principal component of Finance-Activity and Finance-Size.

Finance-Dummy isolates countries that have both underdeveloped financial intermediaries and markets. Specifically, it equals zero if both Private Credit and Value Traded are less than the sample mean and one otherwise.

3.2.2 Indicators of Financial Structure

We also construct measures of the degree to which each country has a market- or bank-based financial system. Since there is not a single accepted definition of financial structure, we use an assortment of different measures to test the robustness of our results. We present the results on four measures of financial structure. Each of these measures is constructed so that higher values indicate more market-based financial systems. Table A2 presents the ranking of countries for the financial structure measures. Demirgüç-Kunt and Levine (1999) examine the relationship between financial structure and a variety of economic, legal and regulatory variables. Along with many findings, they note that higher income countries tend to have more market-oriented financial systems.

would be the same for all industries in this country. Moreover, when we use the turnover ratio, which equals value traded divided by market capitalization, we get the same results. Turnover does not suffer from this price effect because stock prices enter into the numerator and denominator.

Structure-Activity indicates the activity of stock markets relative to the activity of banks and is defined as the log of the ratio of Value Traded and Bank Credit. Bank Credit equals the claims of the banking sector on the private sector as a share of GDP. Compared to Private Credit, we exclude claims of nonbank financial intermediaries to thus focus on the commercial banking sector.

Structure-Size indicates the size of stock markets relative to the size of the banking sector and is defined as the log of the ratio of Market Capitalization and Bank Credit.

Structure-Aggregate combines the previous two measures and is thus a conglomerate indicator of the size and activity of stock markets relative to banks. Specifically, it is the first principal component of Structure-Activity and Structure-Size.

Structure-Dummy is a simple bivariate classification of market- versus bank-based financial systems. Specifically, it equals one if Structure-Aggregate is greater than the sample median and zero otherwise. An economy can be classified as market-based or bank-based only relative to the other countries in the sample, since there is no absolute measure of market- or bank-based financial systems.

These are the most comprehensive measures of financial structure that have been constructed to date. Although they do not directly measure the degree to which banks influence industrial expansion and new firm formation or the ability of markets to stimulate the flow of external finance, the measures – when taken together – provide a measure of the comparative role of banks and markets in the economy. Furthermore, the underlying measures of bank development and stock market liquidity exert a strong influence on economic growth. Thus, in terms of economic growth, the basic measures of bank development and stock market liquidity

are useful measures. Furthermore, Demirgüç -Kunt and Levine (1999) show that countries with strong shareholder rights and high accounting standards tend to have more market-based financial systems. Thus, key legal and regulatory differences match-up with the measures of financial structure that we use to assess the relationship between industrial performance and degree to which countries are bank-based or market-based.

3.2.3 The Legal Environment

We use three indicators of the rights of outside investors and the degree to which these rights are enforced. These data are from La Porta, Lopez-de-Silanes, Shleifer and Vishny (1998).

Creditor is an index of the degree to which the legal codes of the country protect the claims of secured creditors in the case of reorganization or liquidation of a company. It ranges from zero to four and is the sum of four dummy variables that indicate whether (i) the reorganization procedure does not impose an automatic stay on assets, thereby not preventing secured creditors from taking possession of loan collateral, (ii) secured creditors are ranked first in the case of liquidation, (iii) management does not stay in charge of the firm during reorganization, thereby enhancing creditors' power, and (iv) management needs creditors' consent when filing for reorganization. In economies with higher values of Creditor, outside investors have more rights relative to the management and other stakeholders, and should therefore be more willing to provide the external resources that industries need.

Anti-Director is an index of the degree to which the legal codes of the country protect minority shareholder rights. It ranges from zero to six and is the sum of six dummy variables that indicate whether (i) shareholders are allowed to mail their proxy vote to the firm, (ii) shareholders are not required to deposit their shares prior to the General Shareholders' Meeting,

⁹ For evidence on the impact of financial intermediation on growth, see, for example, Levine, Loayza, and Beck (2000). For evidence on the impact of stock markets on growth, see Levine and Zervos (1998) and Rousseau and

(iii) cumulative voting or proportional representation of minorities on the board of directors is allowed, (iv) an oppressed minority mechanism is in place, (v) the minimum percentage of share capital that entitles a shareholder to call for an Extraordinary Shareholders' Meeting is less than or equal to 10 percent, and (vi) shareholders have preemptive rights that can only be waived by a shareholders' vote. In economies with higher values of Anti-Director, minority shareholder are better protected against expropriation by management and large shareholders and should therefore be more willing to provide the external resources that industries need.¹⁰

Rule of Law is an assessment of the law and order tradition of a country that ranges from 10, strong law and order tradition, to 1, weak law and order tradition. This measure was constructed by ICRG and is an average over the period 1982-1990. In countries with a higher law and order tradition, outside investors can more easily enforce their claims and rights and should therefore be more willing to provide external finance.

Table 2 provides descriptive statistics and the correlations for our measures of financial development and structure. There is a large variance in the measures of both financial development and structure. Japan shows the highest value for Finance-Activity and Bangladesh the lowest. Structure-Activity classifies Great Britain as the most market-based system and Bangladesh as the most bank-based system. We also note that most indicators of financial development are positively correlated with the indicators of financial structure, i.e. financially more developed economies tend to have market-based systems. Table A1 lists all measures of financial development, structure and the legal indicators for the countries included in our study.

Wachtel (2000).

¹⁰ Instead of using the sum of the different creditor and minority shareholder rights, we also used principal component indicators, and obtained the same results.

3.3. Industry Growth Rates

Our dependent variable is the average annual growth rate of value added. We use the data obtained by RZ from the *Industrial Statistics Yearbook* database put together by the United Nations Statistical Division (1993). We also use a decomposition of the industry growth rate. Specifically, we consider the growth in the number of establishments and the average size of establishments. ¹¹

Table 3 lists descriptive statistics and correlations between the three different dependent variables. We note that the growth rate in the number of establishments is negatively correlated with the growth rate in the average size of firms. This suggests that industries grow either due to the entry of new firms or due to the expansion of existing firms, but not both.

4. The Results

4.1 Financial Development and Industry Growth

We first present the results of regressions of industry growth rates on the interaction of financial development and external dependence. Our regressions differ from those presented by RZ to the extent that (i) our measures of financial development capture both the effects of financial intermediary and stock market development, and (ii) we include the indicators of financial sector development in logs instead of levels to allow for the nonlinearity in the relationship between financial development and growth illustrated by Levine, Loayza, and Beck (2000). To make our results comparable to those in RZ, we include the "Differential in real"

¹¹ There are no cross-country data available on firms. An establishment is defined as a "unit which engages, under a single ownership or control, in one, or predominantly one, kind of activity at a single location." The growth in the number of establishments (average size of establishment) is defined as the log difference of the number of

growth rate". This indicates how much faster the industry at the 75th percentile of external dependence (Machinery) would have grown compared to the industry at the 25th percentile (Beverages), if Machinery had been located in the country at the 75th percentile of the respective measure of financial development instead of the country at the 25th percentile. Since U.S data are used to calculate our measure of external dependence, the U.S. is dropped from all regressions.

The results in Table 4 indicate a significantly positive interaction of external dependence and overall financial development on industry growth. The coefficients on the interaction terms of all our indicators of financial sector development and external dependence are significantly positive at the five- percent level. We start with the top panel and *Finance-Activity*. The results of the OLS regressions indicate that, for a given industry with a positive external dependence ratio, a higher level of *Finance-Activity* results in a higher growth rate of this industry. To illustrate the significance of this result, consider the growth differential of 2.3 percent. The coefficient estimate thus predicts that Machinery would grow 2.3 percentage points faster than Beverages, if it were located in Malaysia rather than to Greece. The results in columns 2 through 4 indicate that this result is robust to the use of other measures of financial sector development.

The results of the instrumental variable regressions confirm that the link between external dependence, financial development and industry growth is not due to simultaneity bias or reverse causality (Table 4, bottom panel). We report the regression results using TSLS and the legal origin dummies as instruments for financial sector development. The interaction terms with all

establishments (value added in the industry divided by number of establishments) at the beginning and the end of the period.

¹² The growth differential is calculated as follows: 1.553*0.368 (External dependence of machinery minus external dependence of beverages)*3.962(Finance-Activity in Malaysia minus Finance-Activity in Greece).

three indicators of financial development show coefficients that are significant at the ten percent level. The coefficients, however, are of smaller size than in the OLS regressions. The growth differentials of the TSLS regressions are also more in line with the results obtained by RZ. These results therefore indicate that better-developed financial intermediaries and markets ameliorate market frictions and thereby promote the growth of industries that rely more heavily on external finance.

4.2 Financial Structure and Industry Growth

Thus far, the results confirm those in RZ. Industries that depend heavily on external finance grow faster in economies with higher levels of financial development. We will now turn to the question of whether a specific structure of the financial system – bank-based or market-based -- enhances growth of these industries.

The results in Table 5 indicate that the financial structure does not have an independent impact on industrial growth patterns across countries. Although the interaction terms with all measures of financial structure show coefficients that are significant at the five- percent level in the OLS regressions, these results are not confirmed by the instrumental variable regressions.

The results in Table 6 support the financial services view and reject the bank-based and market-based views. Specifically, we find that when controlling for the level of financial development, the interaction of external dependence and financial structure does not have a significant impact on industrial growth patterns across countries. Whereas the interaction terms with all indicators of financial development are significant at the five- percent level, none of the interaction terms with our financial structure measure is significant. These results indicate

¹⁴ In the following we will only present the TSLS results. The OLS regressions yield similar results.

¹³ Since *Finance-Dummy* is a binary variable and because of the findings reported below, we do not attempt to instrument for *Finance-Dummy* using a probit regression in the first stage.

strong evidence in favor of the financial services view and against both the bank-based and market-based view.¹⁵

The results in Table 7 provide support for the legal-based approach. To evaluate the legal-based approach we replace our indicators of financial development with Creditor, Anti-director and $Rule\ of\ Law$. To test for the joint significance of these legal variables, we include an F-test of the three interaction terms. While none of the interaction terms of our financial structure variables is significant, the interaction terms of the three legal variables are jointly significant. The p-values for the individual interaction terms indicate that it is especially the enforcement of investor rights that explains industrial growth patterns across countries.

4.3 Financial Structure and the Sources of Industry Growth

We now decompose the industry growth rates into two components: the growth in the number of establishments and the growth in the average size of establishments. The creation of new establishments is more likely to depend on external funds than the expansion of existing establishments, which can be financed with internal resources. The decomposition of industry growth therefore provides both a robustness test of the previous results and a more detailed exploration of the mechanisms through which financial development and financial structure influence industrial growth patterns across countries.

 $^{^{15}}$ While Levine and Zervos (1998) find evidence that stock markets and banks enhance economic growth through different channels, Levine (2000b) and this paper test the hypothesis that the composition of the financial sector matters for economic growth. These two questions are complementary and not conflicting. For instance, if we had found a statistically significant parameter on financial structure, δ_2 (significantly positive or negative), this would not have invalidated the results obtained by Levine and Zervos (1998). This results would have indicated that market-based systems (or bank-based systems) are more conducive to the growth of financially dependent industries and the emergence of new firms. This result, however, would not necessarily imply that banks (or markets) do not have a positive impact on economic growth.

Again, the results in Table 8 support the financial services view and contradict both the bank-based and market-based views. The results indicate that overall financial development increases the growth in the number of establishments in industries that are dependent on external finance. However, distinguishing by whether a country is bank-based or market-based does not help explain the emergence of new establishments. The results in Table 9 indicate that neither financial development nor structure helps explain the growth rate of the average size of establishments across countries. ¹⁷ This is consistent with findings by RZ that firms depend on external finance during their early years and less during later years.

Table 10 provides evidence consistent with the legal-based view. The legal determinants of financial development can explain industry patterns in the growth in the number of establishments, but not in the growth in the average size of establishments across countries. The interaction terms with financial structure are again insignificant.

In sum, these results indicate that the overall level of financial development and its legal determinants help externally dependent industries grow faster by enabling the start-up of new firms and not through the expansion of existing ones. This is consistent with the Schumpeterian view that financial development enhances economic growth by allowing new firms and projects to develop. These results are also consistent with previous studies that show that financial development enhances economic growth through a better resource allocation and not through capital accumulation [Beck, Levine, and Loayza, 2000].

¹⁶ Alternatively, we could use these legal indicators as instruments to thus extract the exogenous component of financial development explained by these legal rights and their enforcement. The results are similar to the ones reported here.

¹⁷ The results concerning financial development are consistent with the results obtained by Rajan and Zingales (1998).

5. Robustness Tests¹⁸

This section assesses the robustness of the results to alternative measures of financial structure, financial development, and external dependence. First, recognizing that there is not a universally accepted definition of bank-based versus market-based, we decided to isolate those countries with extremely bank-based or market-based systems. Perhaps, very "unbalanced" financial systems are robustly linked with industrial performance, even though continuous measures used thus far are not. We construct three additional variables. *Unbalanced-Bank* equals one if Bank Credit is greater than the sample median and Value Traded is less than the sample median, and zero otherwise. 19 Unbalanced-Market equals one if Value Traded is greater than the sample median and Bank Credit less than the sample median, and zero otherwise.²⁰ Finally, *U nbalanced* equals one if either *U nbalanced Bank* or *U nbalanced Market* equals one, and zero otherwise. The results indicate that classifying countries as having unbalanced financial systems does not help explain industrial growth patterns across countries.

To assess further the robustness of our results, we also constructed measures of financial development and structure that only include the assets privately-owned banks and therefore exclude the assets of state-owned banks. Our findings are robust to the use of credit to the private sector by *privately-owned* financial institutions. Using recently compiled data by La Porta, Lopez-de-Silanes and Shleifer (2000) on the public share in the commercial banking sector, we construct two new measures of (1) credit to the private sector by privately owned deposit money banks and (2) credit to the private sector by privately owned financial

These results are available on request in Appendix B.
 Austria, Chile, Denmark, Finland, and Portugal are classified as having unbalanced bank-based systems.

²⁰ Australia, Brazil, India, New Zealand and Sweden are classified as having unbalanced market-based systems.

intermediaries.²¹ We then recalculate all our indicators of overall financial development and financial structure using these measures to check the robustness of our previous findings.²² These new measures confirm our earlier findings: Neither bank- nor market-based systems have a robust link with the growth patterns of externally dependent industries, new firm creation, or existing firm expansion. The results strongly support the legal-based view. In sum, these additional measures of financial development and structure do not alter the paper's findings.

We also use two measures of financial structure proposed by Demirgüç-Kunt and Maksimovic (2000). Specifically, we regress Value Traded on *Rule of Law*, the British legal origin dummy, the inflation rate and *Anti-Director*.²³ The residuals of this regression reflect the component of stock market development not predicted by the legal and macroeconomic environment. Similarly, we regress Bank Credit on *Rule of Law*, the British legal origin dummy, the inflation rate and *Creditor*. Positive residuals from these two regressions, which we call *Excess-Market* and *Excess-Bank*, indicate stock market and banking sector development that goes beyond the predicted development. We then include interaction terms of external dependence with both residual series in our regressions. A positive coefficient on either interaction term would indicate that externally dependent industries grow faster in countries in which the stock market or banks are larger than predicted by the legal or macroeconomic environment. These alternative measures of financial structure do enter significantly and therefore provide additional support for the legal-based view.

²¹ Specifically, we multiply the measures discussed above by one minus the share of publicly owned commercial banks. We use the average of the government ownership in 1995 and pre-privatization. Both measures are constructed as the percentage of assets of the 10 largest banks in each country owned by the government divided by the total assets of the banking sector. See La Porta, Lopez-de-Silanes, and Shleifer (2000) for details. The correlations between our two new measures and the original ones are 88% and 92%, respectively.

²² Although the public share refers only to commercial banks, we assume that the nonbank financial sector presents a similar ownership structure for each country.

²³ Boyd, Levine, and Smith (2000) show that inflation tends to reduce stock market liquidity and banking sector activity.

The RZ data set contains three alternative measures of external dependence that allow us to test the sensitivity of our results. The three alternative measures of external dependence are significantly correlated with our principal measure of external dependence at the one-percent level, with correlation coefficients being at least 60%. We first use the dependence on external finance of firms that went public during the previous ten years. RZ show that the demand for external finance is highest during the early years of a company. Using a sample of young firms to calculate the dependence on external finance might therefore give a more appropriate picture of the need for external finance. Using the external dependence of young firms does not alter our main result: financial structure does not robustly explain industrial growth patterns, new firm formation, or old firm expansion. When using young firms to define external dependence, there are some specifications in which overall financial development enters insignificantly. However, using the external dependence of young firms lends particularly strong support to the legal-based view of finance and growth.

Our results are also robust to using a measure of external dependence that is calculated for the period 1970-79. RZ suggest that if countries other than the U.S. use older technologies, the external dependence as measured over the 80s might not reflect well the needs for external finance in other countries, especially developing countries. We therefore rerun the regressions using the external dependence measured over the 70s. Since the U.S. was also "more" bankbased in the 70s than in 80s, using this historic measure of external dependence has another advantage. It allows us to test the sensitivity of our results to a bias that might have been introduced by using the external dependence of industries measured for a sample of firms in a market-based economy. Our results are similar to the ones obtained with our principal measure of external finance, as measured over the 80s. There is not a robust link between financial

structure and industrial growth patterns, but overall financial development and the component of overall financial development explained by the legal environment help explain industrial growth patterns, especially the formation of new firms.

Finally, our previous results concerning financial development and financial structure are not due to peculiar characteristics of industries in the U.S. We use the external dependence as calculated for a sample of Canadian firms, which RZ note is the only other country for which firm-level flow of funds are available. We confirm our results concerning financial structure. However, using the Canadian data, we cannot confirm the results concerning the legal-based view and the results on the financial services view are weakened. These results might be partly explained by the fact that we have data for only 27 industries in the Canadian sample, whereas there are at least 36 industries in the text specification. Furthermore, the sample size drops from 1222 to 702. Thus, with some qualifications, the robustness checks confirm the text's main conclusions: (1) industries that are heavily dependent on external finance do not grow faster in bank-based or market-based financial system, (2) externally dependent industries do, however, tend to grow faster in countries with better-developed financial systems and especially in economies that efficiently project the legal rights of outside investors, and (3) overall financial development and the legal protection of investors stimulates industry growth primarily by facilitating new firm formation.

6. Conclusions

This paper examined the following questions: Do industries that depend heavily on external finance grow faster in bank-based or market-based systems? Are new firms more likely to form in a bank-based or market-based financial system? Alternatively, is it the overall level of

financial development or the legal system that explains industrial growth patterns and the emergence of new firms across countries?

The results do not provide support for either the bank-based or the market-based view. Measuring whether a country is bank-based or market-based does not help explain industrial growth patterns. The results do indicate, however, that industries that are heavy users of external finance grow faster in countries with higher overall levels of financial development and in countries that rigorously protect the rights of outside investors. Moreover, the findings show that the overall level of financial development along with strong creditor rights, shareholder rights, and contract enforcement mechanisms foster new firm formation. Together, these findings provide support for the financial-services and legal-based views.

REFERENCES

- Allen, Franklin and Gale, Douglas. <u>Comparing Financial Systems</u>. Cambridge, MA: MIT Press, 1999.
- Allen, Franklin. "Stock Markets and Resource Allocation", in: C. Mayer and X. Vives (eds.): Capital Markets and Financial Intermediation, 1993, Cambridge University Press, Cambridge.
- Beck, Thorsten; Demirgüç-Kunt, Asli; Levine, Ross. "A New Database on Financial Development and Structure", World Bank Economic Review, 2000 forthcoming.
- Beck, Thorsten; Levine, Ross; Loayza, Norman. "Finance and the Sources of Growth," <u>Journal</u> of Financial Economics, 2000, forthcoming.
- Boot, Arnoud W.A.; Greenbaum, Stuart J.; Thakor, Anjan V. "Reputation and Discretion in Financial Contracting," <u>American Economic Review</u>, 1993, 83, 1165-83.
- Boot, Arnoud W.A. and Thakor, Anjan. "Financial System Architecture," <u>Review of Financial Studies</u>, Fall 1997, 10(3), pp. 693-733.
- Boot, Arnoud W.A. and Thakor, Anjan V. "Can Relationship Banking Survive Competition?" <u>Journal of Finance</u>, April 2000, 55(2), 679-713.
- Boyd, John H. and Prescott, Edward C. "Financial Intermediary-Coalitions," <u>Journal of Economics Theory</u>, April 1986, <u>38</u>(2), pp. 211-32.
- Boyd, John H. and Smith, Bruce D. "The Evolution of Debt and Equity Markets in Economic Development," <u>Economic Theory</u>, 12(3), 1998, pp. 519-60.
- Boyd, John H.; Levine, Ross; and Smith, Bruce D. "The Impact of Inflation on Financial Sector Performance," <u>Carlson School of Management (University of Minnesota), mimeo, 2000.</u>
- Cetorelli, Nicola and Gamberra, Michele. "Banking Market Structure, Financial Dependence and Growth: International Evidence from Industry Data", Federal Reserve Bank of Chicago, mimeo, 2000.
- Demirgüç-Kunt, Asli and Levine, Ross. "Financial Structures Across Countries: Stylized Facts" Washington, D.C.: World Bank Policy Research Working Paper 2143, 1999.
- Demirgüç-Kunt, Asli and Maksimovic, Vojislav. "Law, Finance, and Firm Growth," <u>Journal of Finance</u>, December 1998, <u>53</u>(6), pp.2107-2137.
- Demirgüç-Kunt, Asli and Maksimovic, Vojislav. "Funding Growth in Bank-Based and Market-Based Financial Systems: Evidence from Firm Level Data, mimeo, 2000.

- Diamond, Douglas W. "Financial Intermediation and Delegated Monitoring," <u>Review of Economic Studies</u>, July 1984, <u>51</u>(3), pp. 393-414.
- Gerschenkron, Alexander. <u>Economic Backwardness in Historical Perspective</u>, A <u>Book of Essays</u>. Cambridge, MA: Harvard University Press, 1962,
- Grossman, Sanford, and Hart, Oliver. "Takeover Bids, the Free-Rider Problem, and the Theory of the Corporation," <u>Bell Journal of Economics</u>, 1980, <u>11</u>, 42-64.
- Hellwig, Martin. "Banking, Financial Intermediation, and Corporate Finance," in: A. Giovanni and C. Mayer (eds.) European Financial Intermediation, 1991, Cambridge University Press, Cambridge, pp.35-63.
- Hellwig, Martin. "On the Economics and Politics of Corporate Finance and Corporate Control," working paper, University of Mannheim, 1998.
- Hoshi, Takeo, Kashyap, Anil, Scharfstein, David. "Corporate Structure, Liquidity, and Investment: Evidence from Japanese Industrial Groups," <u>0 uarterly Journal of Economics</u>, 106, 1991, pp. 33-60.
- Huybens, Elisabeth, and Smith, Bruce, 1999, Inflation, Financial Markets, and Long-Run Real Activity, <u>Journal of Monetary Economics</u>, 43, 283-315.
- King, Robert G. and Levine, Ross. "Finance, Entrepreneurship, and Growth: Theory and Evidence," <u>Journal of Monetary Economics</u>, December 1993, <u>32</u>(3), pp. 513-42.
- La Porta, Rafael; Lopez-de-Silanes, Florencio; and Shleifer, Andrei. "Government Ownership of Commercial Banks", NBER Working Paper7620, March 2000.
- La Porta, Rafael; Lopez-de-Silanes, Florencio; Shleifer, Andrei; and Vishny, Robert W. "Investor Protection and Corporate Governance," mimeo, 1999b.
- La Porta, Rafael; Lopez-de-Silanes, Florencio; Shleifer, Andrei; and Vishny, Robert W. "The Quality of Government," <u>Journal of Law, Economics, and Organization</u>, 1999a, <u>15</u>(1), pp. 222-279.
- La Porta, Rafael; Lopez-de-Silanes, Florencio; Shleifer, Andrei; and Vishny, Robert W. "Law and Finance," <u>Journal of Political Economy</u>, 1998, <u>106</u>(6), pp. 1113-1155
- La Porta, Rafael; Lopez-de-Silanes, Florencio; Shleifer, Andrei; and Vishny, Robert W. "Legal Determinants of External Finance," <u>Journal of Finance</u>, July 1997, <u>52</u>(3), pp. 1131-1150.
- Levine, Ross. "Financial Development and Economic Growth: Views and Agenda," <u>Journal of Economic Literature</u>, June 1997, 3592), pp. 688-726.

- Levine, Ross. "The Legal Environment, Banks, and Long-Run Economic Growth," <u>Journal of Money, Credit, and Banking</u>, August 1998.
- Levine, Ross. "Law, Finance, and Economic Growth", <u>Journal of Financial Intermediation</u>, 1999, <u>8</u>(1/2), pp. 36-67.
- Levine, Ross. "Napoleon, Bourses, and Growth: With A Focus on Latin America," in <u>Market Augmenting Government</u>, Eds. Omar Azfar and Charles Cadwell. Washington, D.C.: IRIS, 2000a, *forthcoming*.
- Levine, Ross. "Bank-Based or Market-Based Financial Systems: Which is Better?," Carlson School of Management (University of Minnesota), mimeo 2000b.
- Levine, Ross; Loayza, Norman; Beck, Thorsten. "Financial Intermediation and Growth: Causality and Causes," <u>Journal of Monetary Economics</u>, 2000, forthcoming.
- Mork, Randall and Nakkamura, Masao. "Banks and Corporate Control in Japan," <u>Journal of Finance</u>, 1999, 54, 319-40.
- Petersen, Mitchell; and Rajan, Raghuram G. "The Effect of Credit Market Competition on Lending Relationships", <u>0 uarterly Journal of Economics</u>, 1995, <u>110</u>, pp 407-443.
- Rajan, Raghuram G. "Insiders and Outsiders: The Choice Between Informed and Arms Length Debt," Journal of Finance, September 1992, 47(4), pp. 1367-1400.
- Rajan, Raghuram G. and Zingales, Luigi. "Financial Dependence and Growth," <u>American Economic Review</u>, June 1998.
- Rajan, Raghuram G. and Zingales, Luigi. "Financial Systems, Industrial Structure, and Growth, mimeo 1999.
- Ramakrishnan, Ram and Thakor, Anjan V. "Information Reliability and a Theory of Financial Intermediation," Review of Economic Studies, 1984, 51, 415-432
- Reynolds, Thomas H. and Flores, Arturo A. <u>Foreign Law Current Sources of Codes and Legislation in Jurisdictions of the World</u>, Fred B. Rothman & Co. (Littleton Colorado), (AALL Publication Series No. 33). 1996.
- Rousseau, Peter L. and Wachtel, Paul. "Equity Markets and Growth: Cross-Country Evidence on Timing and Outcomes, 1980-1995, <u>Journal of Business and Finance</u>, 2000, forthcoming.
- Shleifer, Andrei and Vishny, Robert W. "Large Shareholders and Corporate Control," <u>Journal of Political Economy</u>, June 1986, <u>96</u>(3), pp. 461-88.
- Stiglitz, Joseph E. "Credit Markets and the Control of Capital," <u>Journal of Money, Credit and Banking</u>, May 1985, <u>17</u>(2), pp. 133-52.

- Stulz, Rene M. "Financial Structure, Corporate Finance, and Economic Growth," 2000, Ohio State University mimeo.
- Weinstein, David E. and Y afeh, Y ishay. "On the Costs of a Bank-Centered Financial System: Evidence from the Changing Main Bank Relations in Japan," <u>Journal of Finance</u>, 1998, 53(2), pp. 635-672.
- Wenger, Ekkehard and Kaserer, Christoph. "The German System of Corporate Governance: A Model Which Should Not Be Imitated," in "in Competition and Convergence in Financial Markets: The German and Anglo-American Models, Eds: Stanley W. Black and Mathias Moersch, New Y ork: North -Holland Press, 1998, pp. 41-78.
- Winton, Andrew. "Delegated Monitoring and Bank Structure in a Finite Economy," <u>Journal of Financial Intermediation</u>, 1995, 4, 158-87.

Table 1: Measures of External Dependence Across Industries

314 Tobacco	ISIC code Industrial Sector	All U.S.	Young U.S. companies, 80s	All U.S. companies, 70s	Canadian companies, 80s
361 Pottery 323 Leather 324 Spinning 324 Footwear 324 Footwear 326 Nonferrous metal 327 Nonferrous metal 327 Nonferrous metal 328 Petroleum refineries 328 Nonmetal mineral products 329 Nonmetal mineral products 320 Nonferrous metal 320 Nonmetal mineral products 320 Nonmetal mineral products 321 Sperroleum refineries 322 Nonferrous metal 323 Petroleum refineries 324 Potoleum refineries 325 Nonmetal mineral products 326 Nonmetal mineral products 327 Nonferrous Monte					
361 Pottery 323 Leather 324 Spinning 324 Footwear 324 Footwear 326 Nonferrous metal 327 Nonferrous metal 327 Nonferrous metal 328 Petroleum refineries 328 Nonmetal mineral products 329 Nonmetal mineral products 320 Nonferrous metal 320 Nonmetal mineral products 320 Nonmetal mineral products 321 Sperroleum refineries 322 Nonferrous metal 323 Petroleum refineries 324 Potoleum refineries 325 Nonmetal mineral products 326 Nonmetal mineral products 327 Nonferrous Monte	314 Tobacco	-0.45		-0.13	-0.59
323 Leather 3211 Spinning -0.09 324 Footwear 372 Nonferrous metal 30.01 325 Apparel 30.03 326 Apparel 30.03 327 327 328 Petroleum refineries 30.04 329 Nonmetal mineral products 30.06 30.09 311 Iron and steel 311 Food products 311 Food products 311 Pulp, paper 311 Paper 312 Apparel 323 Synthetic resins 324 Petroleum and publishing 325 Other chemicals 326 Machinery 327 328 Machinery 328 Petroleum and coal products 329 320 Apparel 30.00 30.07 320 30.09 30.09 30.09 30.09 30.09 30.09 30.09 30.09 30.09 30.09 30.09 30.00 30.09 30.09 30.00 30.09 30.00 30.09 30.00 30.09 30.00 30.09 30.00 30.09 30.00 30.09 30.00 30.09 30.00 30.09 30.00 30.00 30.09 30.00 30.09 30.00 30		-0.15	-0.41	-0.45	
3211 Spinning -0.09 -0.05 324 Footwear -0.08 -0.06 -0.26 -0.09 -0.09 324 Footwear -0.09 -0.09 -0.09 322 Apparel -0.01 0.46 0.19 -0.09 322 Apparel -0.03 0.27 0.03 353 Petroleum refineries -0.04 0.85 0.06 -0.01 369 Nonmetal mineral products 0.06 -0.03 0.09 -0.12 313 Beverages 0.08 0.63 -0.06 0.59 371 Iron and steel -0.09 0.26 -0.01 0.22 311 Food products 0.14 0.86 0.06 0.11 3411 Pulp, paper 0.15 0.22 0.08 0.15 3513 Synthetic resins 0.16 0.79 0.03 -0.24 341 Paper and paper products 0.18 0.57 -0.01 -0.15 342 Printing and publishing 0.20 0.60 -0.01 0.38 352 Other chemicals 0.22 1.35 -0.07 -0.80 355 Rubber products 0.23 0.50 0.07 325 Furniture 0.24 0.68 0.16 3511 Basic industrial goods excl. fertilizers 0.24 0.68 0.16 3511 Basic industrial goods excl. fertilizers 0.25 0.79 0.21 0.38 331 Wood products 0.28 0.34 0.28 0.30 334 Transportation equipment 0.31 0.58 0.23 0.79 354 Petroleum and coal products 0.33 -0.26 -0.21 384 Motor vehicles 0.39 0.76 0.13 0.29 321 Textile 0.40 0.66 -0.04 0.57 332 Machinery 0.45 0.75 0.16 0.34 3841 Ships 0.46 1.05 0.15 390 Other industries 0.47 0.80 0.12 0.69 362 Glass 0.53 0.50 0.97 0.56 0.53 383 Electric machinery 0.77 1.22 0.26 0.75 385 Professional and scientific goods 1.04 1.35 0.411 1.09 3825 Office and computing products 1.06 1.16 0.54 1.19	•	-0.14	-1.53	-0.04	
324 Footwear -0.08 0.65 -0.26 372 Nonferrous metal 0.01 0.46 0.19 -0.09 322 Apparel 0.03 0.27 0.03 353 Petroleum refineries 0.04 0.85 0.06 -0.01 369 Nonmetal mineral products 0.06 -0.03 0.09 -0.12 313 Beverages 0.08 0.63 -0.06 0.59 371 Iron and steel 0.09 0.26 -0.01 0.22 311 Food products 0.14 0.66 0.06 0.11 3411 Pulp, paper 0.15 0.22 0.08 0.15 3513 Synthetic resins 0.16 0.79 0.03 -0.24 341 Paper and paper products 0.18 0.57 -0.01 -0.15 342 Printing and publishing 0.20 0.60 -0.01 0.38 352 Other chemicals 0.22 1.35 -0.07 -0.80 355 Rubber products 0.23 0.50 0.07 -0.80 351 Basic industrial goods e	3211 Spinning	-0.09		0.05	
322 Apparel 0.03 0.27 0.03 363 Petroleum refineries 0.04 0.85 0.06 -0.01 369 Nonmetal mineral products 0.06 -0.03 0.09 -0.12 313 Beverages 0.08 0.63 -0.06 0.59 371 Iron and steel 0.09 0.26 -0.01 0.22 311 Food products 0.14 0.66 0.06 0.11 3411 Pulp, paper 0.15 0.22 0.08 0.15 3513 Synthetic resins 0.16 0.79 0.03 -0.24 341 Paper and paper products 0.18 0.57 -0.01 -0.15 342 Printing and publishing 0.20 0.60 -0.01 0.38 352 Other chemicals 0.22 1.35 -0.07 -0.80 355 Rubber products 0.23 0.50 0.07 331 Metai products 0.24 0.68 0.16 381 Metai products 0.24 0.87 0.17 0.61 381 Wood products 0.28 0.34 0.28 0.30 384 Transportation equipment		-0.08	0.65	-0.26	
353 Petroleum refineries 0.04 0.85 0.06 -0.01 369 Nonmetal mineral products 0.06 -0.03 0.09 -0.12 313 Beverages 0.08 0.63 -0.06 0.59 371 Iron and steel 0.09 0.26 -0.01 0.22 311 Food products 0.14 0.66 0.06 0.11 3411 Pulp, paper 0.15 0.22 0.08 0.15 3513 Synthetic resins 0.16 0.79 0.03 -0.24 341 Paper and paper products 0.18 0.57 -0.01 -0.15 342 Printing and publishing 0.20 0.60 -0.01 0.38 352 Other chemicals 0.22 1.35 -0.07 -0.80 355 Rubber products 0.23 0.50 0.07 322 Furniture 0.24 0.68 0.16 381 Metal products 0.24 0.87 0.17 0.61 3511 Basic industrial goods excl. fertilizers 0.25 0.79 0.21 0.38 <td< td=""><td>372 Nonferrous metal</td><td>0.01</td><td>0.46</td><td>0.19</td><td>-0.09</td></td<>	372 Nonferrous metal	0.01	0.46	0.19	-0.09
353 Petroleum refineries 0.04 0.85 0.06 -0.01 369 Nonmetal mineral products 0.06 -0.03 0.09 -0.12 313 Beverages 0.08 0.63 -0.06 0.59 371 Iron and steel 0.09 0.26 -0.01 0.22 311 Food products 0.14 0.66 0.06 0.11 3411 Pulp, paper 0.15 0.22 0.08 0.15 3513 Synthetic resins 0.16 0.79 0.03 -0.24 341 Paper and paper products 0.18 0.57 -0.01 -0.15 342 Printing and publishing 0.20 0.60 -0.01 0.38 352 Other chemicals 0.22 1.35 -0.07 -0.80 355 Rubber products 0.23 0.50 0.07 322 Furniture 0.24 0.68 0.16 381 Metal products 0.24 0.87 0.17 0.61 3511 Basic industrial goods excl. fertilizers 0.25 0.79 0.21 0.38 <td< td=""><td>322 Apparel</td><td>0.03</td><td>0.27</td><td>0.03</td><td></td></td<>	322 Apparel	0.03	0.27	0.03	
369 Nonmetal mineral products 0.06 -0.03 0.09 -0.12 313 Beverages 0.08 0.63 -0.06 0.59 371 Iron and steel 0.09 0.26 -0.01 0.22 311 Food products 0.14 0.66 0.06 0.11 3411 Pulp, paper 0.15 0.22 0.08 0.15 3513 Synthetic resins 0.16 0.79 0.03 -0.24 341 Paper and paper products 0.18 0.57 -0.01 -0.15 342 Printing and publishing 0.20 0.60 -0.01 -0.38 352 Other chemicals 0.22 1.35 -0.07 -0.80 355 Rubber products 0.23 0.50 0.07 332 Furniture 0.24 0.68 0.16 381 Metal products 0.24 0.87 0.17 0.61 3511 Basic industrial goods excl. fertilizers 0.25 0.79 0.21 0.38 331 Wood products 0.28 0.34 0.28 0.30 344 Transportation equipment 0.31 0.58 0.23 0.79		0.04	0.85	0.06	-0.01
313 Beverages 0.08 0.63 -0.06 0.59 371 Iron and steel 0.09 0.26 -0.01 0.22 311 Food products 0.14 0.66 0.06 0.11 3411 Pulp, paper 0.15 0.22 0.08 0.15 3513 Synthetic resins 0.16 0.79 0.03 -0.24 341 Paper and paper products 0.18 0.57 -0.01 -0.15 342 Printing and publishing 0.20 0.60 -0.01 0.38 352 Other chemicals 0.22 1.35 -0.07 -0.80 355 Rubber products 0.23 0.50 0.07 332 Furniture 0.24 0.68 0.16 381 Metal products 0.24 0.87 0.17 0.61 3511 Basic industrial goods excl. fertilizers 0.25 0.79 0.21 0.38 331 Wood products 0.28 0.34 0.28 0.30 384 Transportation equipment 0.31 0.58 0.23 0.79 342 Petroleum and coal products 0.39 0.76 0.13 0.29		0.06	-0.03	0.09	-0.12
311 Food products 0.14 0.66 0.06 0.11 3411 Pulp, paper 0.15 0.22 0.08 0.15 3513 Synthetic resins 0.16 0.79 0.03 -0.24 341 Paper and paper products 0.18 0.57 -0.01 -0.15 342 Printing and publishing 0.20 0.60 -0.01 0.38 352 Other chemicals 0.22 1.35 -0.07 -0.80 355 Rubber products 0.23 0.50 0.07 332 Furniture 0.24 0.68 0.16 381 Metal products 0.24 0.87 0.17 0.61 3511 Basic industrial goods excl. fertilizers 0.25 0.79 0.21 0.38 331 Wood products 0.28 0.34 0.28 0.30 384 Transportation equipment 0.31 0.58 0.23 0.79 354 Petroleum and coal products 0.33 -0.26 -0.21 3843 Motor vehicles 0.39 0.76 0.13 0.29 321 Textile 0.40 0.66 -0.04 0.57 382 Machiner	•	0.08	0.63	-0.06	0.59
3411 Pulp, paper 0.15 0.22 0.08 0.15 3513 Synthetic resins 0.16 0.79 0.03 -0.24 341 Paper and paper products 0.18 0.57 -0.01 -0.15 342 Printing and publishing 0.20 0.60 -0.01 0.38 352 Other chemicals 0.22 1.35 -0.07 -0.80 355 Rubber products 0.23 0.50 0.07 332 Furniture 0.24 0.68 0.16 381 Metal products 0.24 0.87 0.17 0.61 3511 Basic industrial goods excl. fertilizers 0.25 0.79 0.21 0.38 331 Wood products 0.28 0.34 0.28 0.30 384 Transportation equipment 0.31 0.58 0.23 0.79 354 Petroleum and coal products 0.33 -0.26 -0.21 3843 Motor vehicles 0.39 0.76 0.13 0.29 321 Textile 0.40 0.66 -0.04 0.57 382 Machinery 0.45 0.75 0.16 0.34 3841 Ships	371 Iron and steel	0.09	0.26	-0.01	0.22
3411 Pulp, paper 0.15 0.22 0.08 0.15 3513 Synthetic resins 0.16 0.79 0.03 -0.24 341 Paper and paper products 0.18 0.57 -0.01 -0.15 342 Printing and publishing 0.20 0.60 -0.01 0.38 352 Other chemicals 0.22 1.35 -0.07 -0.80 355 Rubber products 0.23 0.50 0.07 332 Furniture 0.24 0.68 0.16 381 Metal products 0.24 0.87 0.17 0.61 3511 Basic industrial goods excl. fertilizers 0.25 0.79 0.21 0.38 331 Wood products 0.28 0.34 0.28 0.30 384 Transportation equipment 0.31 0.58 0.23 0.79 354 Petroleum and coal products 0.33 -0.26 -0.21 3843 Motor vehicles 0.39 0.76 0.13 0.29 321 Textile 0.40 0.66 -0.04 0.57 382 Machinery 0.45 0.75 0.16 0.34 3841 Ships	311 Food products	0.14	0.66	0.06	0.11
3513 Synthetic resins 0.16 0.79 0.03 -0.24 341 Paper and paper products 0.18 0.57 -0.01 -0.15 342 Printing and publishing 0.20 0.60 -0.01 0.38 352 Other chemicals 0.22 1.35 -0.07 -0.80 355 Rubber products 0.23 0.50 0.07 325 Rubber products 0.23 0.50 0.07 381 Metal products 0.24 0.68 0.16 381 Metal products 0.24 0.87 0.17 0.61 351 Basic industrial goods excl. fertilizers 0.25 0.79 0.21 0.38 331 Wood products 0.28 0.34 0.28 0.30 344 Transportation equipment 0.31 0.58 0.23 0.79 354 Petroleum and coal products 0.33 -0.26 -0.21 3843 Motor vehicles 0.39 0.76 0.13 0.29 321 Textile 0.40 0.66 -0.04 0.57 382 Machinery 0.45 0.75 0.16 0.34 3841 Ships 0.4		0.15	0.22	0.08	0.15
342 Printing and publishing 0.20 0.60 -0.01 0.38 352 Other chemicals 0.22 1.35 -0.07 -0.80 355 Rubber products 0.23 0.50 0.07 332 Furniture 0.24 0.68 0.16 381 Metal products 0.24 0.87 0.17 0.61 3511 Basic industrial goods excl. fertilizers 0.25 0.79 0.21 0.38 331 Wood products 0.28 0.34 0.28 0.30 384 Transportation equipment 0.31 0.58 0.23 0.79 354 Petroleum and coal products 0.33 -0.26 -0.21 3843 Motor vehicles 0.39 0.76 0.13 0.29 321 Textile 0.40 0.66 -0.04 0.57 382 Machinery 0.45 0.75 0.16 0.34 3841 Ships 0.46 1.05 0.15 390 Other industries 0.47 0.80 0.12 0.69 362 Glass 0.53 1.52 0.07 0.56 383 Electric machinery 0.77 1.22 <td></td> <td>0.16</td> <td>0.79</td> <td>0.03</td> <td>-0.24</td>		0.16	0.79	0.03	-0.24
342 Printing and publishing 0.20 0.60 -0.01 0.38 352 Other chemicals 0.22 1.35 -0.07 -0.80 355 Rubber products 0.23 0.50 0.07 332 Furniture 0.24 0.68 0.16 381 Metal products 0.24 0.87 0.17 0.61 3511 Basic industrial goods excl. fertilizers 0.25 0.79 0.21 0.38 331 Wood products 0.28 0.34 0.28 0.30 384 Transportation equipment 0.31 0.58 0.23 0.79 354 Petroleum and coal products 0.33 -0.26 -0.21 3843 Motor vehicles 0.39 0.76 0.13 0.29 321 Textile 0.40 0.66 -0.04 0.57 382 Machinery 0.45 0.75 0.16 0.34 3841 Ships 0.46 1.05 0.15 390 Other industries 0.47 0.80 0.12 0.69 362 Glass 0.53 1.52 0.07 0.56 383 Electric machinery 0.77 1.22 <td>341 Paper and paper products</td> <td>0.18</td> <td>0.57</td> <td>-0.01</td> <td>-0.15</td>	341 Paper and paper products	0.18	0.57	-0.01	-0.15
355 Rubber products 0.23 0.50 0.07 332 Furniture 0.24 0.68 0.16 381 Metal products 0.24 0.87 0.17 0.61 3511 Basic industrial goods excl. fertilizers 0.25 0.79 0.21 0.38 331 Wood products 0.28 0.34 0.28 0.30 384 Transportation equipment 0.31 0.58 0.23 0.79 354 Petroleum and coal products 0.33 -0.26 -0.21 3843 Motor vehicles 0.39 0.76 0.13 0.29 321 Textile 0.40 0.66 -0.04 0.57 382 Machinery 0.45 0.75 0.16 0.34 3841 Ships 0.46 1.05 0.15 390 Other industries 0.47 0.80 0.12 0.69 362 Glass 0.53 1.52 0.07 0.56 383 Electric machinery 0.77 1.22 0.26 0.75 385 Professional and scientific goods 0.96 1.63 0.40 0.51 3825 Office and computing products 1.06<		0.20	0.60	-0.01	0.38
332 Furniture 0.24 0.68 0.16 381 Metal products 0.24 0.87 0.17 0.61 3511 Basic industrial goods excl. fertilizers 0.25 0.79 0.21 0.38 331 Wood products 0.28 0.34 0.28 0.30 384 Transportation equipment 0.31 0.58 0.23 0.79 354 Petroleum and coal products 0.33 -0.26 -0.21 3843 Motor vehicles 0.39 0.76 0.13 0.29 321 Textile 0.40 0.66 -0.04 0.57 382 Machinery 0.45 0.75 0.16 0.34 3841 Ships 0.46 1.05 0.15 390 Other industries 0.47 0.80 0.12 0.69 362 Glass 0.53 1.52 0.07 0.56 383 Electric machinery 0.77 1.22 0.26 0.75 385 Professional and scientific goods 0.96 1.63 0.40 0.51 3825 Office and computing products 1.06 1.16 0.54 1.19	352 Other chemicals	0.22	1.35	-0.07	-0.80
381 Metal products 0.24 0.87 0.17 0.61 3511 Basic industrial goods excl. fertilizers 0.25 0.79 0.21 0.38 331 Wood products 0.28 0.34 0.28 0.30 384 Transportation equipment 0.31 0.58 0.23 0.79 354 Petroleum and coal products 0.33 -0.26 -0.21 3843 Motor vehicles 0.39 0.76 0.13 0.29 321 Textile 0.40 0.66 -0.04 0.57 382 Machinery 0.45 0.75 0.16 0.34 3841 Ships 0.46 1.05 0.15 390 Other industries 0.47 0.80 0.12 0.69 362 Glass 0.53 1.52 0.07 0.56 383 Electric machinery 0.77 1.22 0.26 0.75 385 Professional and scientific goods 0.96 1.63 0.40 0.51 3825 Office and computing products 1.06 1.16 0.54 1.19	355 Rubber products	0.23	0.50	0.07	
3511 Basic industrial goods excl. fertilizers 0.25 0.79 0.21 0.38 331 Wood products 0.28 0.34 0.28 0.30 384 Transportation equipment 0.31 0.58 0.23 0.79 354 Petroleum and coal products 0.33 -0.26 -0.21 3843 Motor vehicles 0.39 0.76 0.13 0.29 321 Textile 0.40 0.66 -0.04 0.57 382 Machinery 0.45 0.75 0.16 0.34 3841 Ships 0.46 1.05 0.15 390 Other industries 0.47 0.80 0.12 0.69 362 Glass 0.53 1.52 0.07 0.56 383 Electric machinery 0.77 1.22 0.26 0.75 385 Professional and scientific goods 0.96 1.63 0.40 0.51 3825 Office and computing products 1.06 1.16 0.54 1.19	332 Furniture	0.24	0.68	0.16	
331 Wood products 0.28 0.34 0.28 0.30 384 Transportation equipment 0.31 0.58 0.23 0.79 354 Petroleum and coal products 0.33 -0.26 -0.21 3843 Motor vehicles 0.39 0.76 0.13 0.29 321 Textile 0.40 0.66 -0.04 0.57 382 Machinery 0.45 0.75 0.16 0.34 3841 Ships 0.46 1.05 0.15 390 Other industries 0.47 0.80 0.12 0.69 362 Glass 0.53 1.52 0.07 0.56 383 Electric machinery 0.77 1.22 0.26 0.75 385 Professional and scientific goods 0.96 1.63 0.40 0.51 3832 Radios 1.04 1.35 0.41 1.09 3825 Office and computing products 1.06 1.16 0.54 1.19	381 Metal products	0.24	0.87	0.17	0.61
384 Transportation equipment 0.31 0.58 0.23 0.79 354 Petroleum and coal products 0.33 -0.26 -0.21 3843 Motor vehicles 0.39 0.76 0.13 0.29 321 Textile 0.40 0.66 -0.04 0.57 382 Machinery 0.45 0.75 0.16 0.34 3841 Ships 0.46 1.05 0.15 390 Other industries 0.47 0.80 0.12 0.69 362 Glass 0.53 1.52 0.07 0.56 383 Electric machinery 0.77 1.22 0.26 0.75 385 Professional and scientific goods 0.96 1.63 0.40 0.51 3832 Radios 1.04 1.35 0.41 1.09 3825 Office and computing products 1.06 1.16 0.54 1.19	3511 Basic industrial goods excl. fertilizers	0.25	0.79	0.21	0.38
354 Petroleum and coal products 0.33 -0.26 -0.21 3843 Motor vehicles 0.39 0.76 0.13 0.29 321 Textile 0.40 0.66 -0.04 0.57 382 Machinery 0.45 0.75 0.16 0.34 3841 Ships 0.46 1.05 0.15 390 Other industries 0.47 0.80 0.12 0.69 362 Glass 0.53 1.52 0.07 0.56 383 Electric machinery 0.77 1.22 0.26 0.75 385 Professional and scientific goods 0.96 1.63 0.40 0.51 3832 Radios 1.04 1.35 0.41 1.09 3825 Office and computing products 1.06 1.16 0.54 1.19	331 Wood products	0.28	0.34	0.28	0.30
3843 Motor vehicles 0.39 0.76 0.13 0.29 321 Textile 0.40 0.66 -0.04 0.57 382 Machinery 0.45 0.75 0.16 0.34 3841 Ships 0.46 1.05 0.15 390 Other industries 0.47 0.80 0.12 0.69 362 Glass 0.53 1.52 0.07 0.56 383 Electric machinery 0.77 1.22 0.26 0.75 385 Professional and scientific goods 0.96 1.63 0.40 0.51 3832 Radios 1.04 1.35 0.41 1.09 3825 Office and computing products 1.06 1.16 0.54 1.19	384 Transportation equipment	0.31	0.58	0.23	0.79
321 Textile 0.40 0.66 -0.04 0.57 382 Machinery 0.45 0.75 0.16 0.34 3841 Ships 0.46 1.05 0.15 390 Other industries 0.47 0.80 0.12 0.69 362 Glass 0.53 1.52 0.07 0.56 383 Electric machinery 0.77 1.22 0.26 0.75 385 Professional and scientific goods 0.96 1.63 0.40 0.51 3832 Radios 1.04 1.35 0.41 1.09 3825 Office and computing products 1.06 1.16 0.54 1.19	354 Petroleum and coal products		-0.26	-0.21	
382 Machinery 0.45 0.75 0.16 0.34 3841 Ships 0.46 1.05 0.15 390 Other industries 0.47 0.80 0.12 0.69 362 Glass 0.53 1.52 0.07 0.56 383 Electric machinery 0.77 1.22 0.26 0.75 385 Professional and scientific goods 0.96 1.63 0.40 0.51 3832 Radios 1.04 1.35 0.41 1.09 3825 Office and computing products 1.06 1.16 0.54 1.19	3843 Motor vehicles	0.39	0.76	0.13	0.29
3841 Ships 0.46 1.05 0.15 390 Other industries 0.47 0.80 0.12 0.69 362 Glass 0.53 1.52 0.07 0.56 383 Electric machinery 0.77 1.22 0.26 0.75 385 Professional and scientific goods 0.96 1.63 0.40 0.51 3832 Radios 1.04 1.35 0.41 1.09 3825 Office and computing products 1.06 1.16 0.54 1.19	321 Textile	0.40	0.66	-0.04	0.57
390 Other industries 0.47 0.80 0.12 0.69 362 Glass 0.53 1.52 0.07 0.56 383 Electric machinery 0.77 1.22 0.26 0.75 385 Professional and scientific goods 0.96 1.63 0.40 0.51 3832 Radios 1.04 1.35 0.41 1.09 3825 Office and computing products 1.06 1.16 0.54 1.19	382 Machinery	0.45	0.75	0.16	0.34
362 Glass 0.53 1.52 0.07 0.56 383 Electric machinery 0.77 1.22 0.26 0.75 385 Professional and scientific goods 0.96 1.63 0.40 0.51 3832 Radios 1.04 1.35 0.41 1.09 3825 Office and computing products 1.06 1.16 0.54 1.19		0.46	1.05	0.15	
383 Electric machinery 0.77 1.22 0.26 0.75 385 Professional and scientific goods 0.96 1.63 0.40 0.51 3832 Radios 1.04 1.35 0.41 1.09 3825 Office and computing products 1.06 1.16 0.54 1.19	390 Other industries	0.47	0.80	0.12	0.69
385 Professional and scientific goods 0.96 1.63 0.40 0.51 3832 Radios 1.04 1.35 0.41 1.09 3825 Office and computing products 1.06 1.16 0.54 1.19			1.52	0.07	0.56
3832 Radios 1.04 1.35 0.41 1.09 3825 Office and computing products 1.06 1.16 0.54 1.19				0.26	
3825 Office and computing products 1.06 1.16 0.54 1.19	385 Professional and scientific goods	0.96	1.63	0.40	0.51
		1.04	1.35	0.41	1.09
				0.54	
·	356 Plastic products	1.14	1.14		0.48
3522 Drugs 1.49 2.06 0.09 3.51	3522 Drugs	1.49	2.06	0.09	3.51

External dependence is defined as capital expenditures (Compustat # 128) minus cash flow from operations divided by capital expenditures. Cash flow from operations is broadly defined as the sum of Compustat funds from operations(items # 110), decreases in inventories, decreases in receivables, and increases in payables.

Source: Rajan and Zingales (1998)

Table 2: Financial Development and Structure Across Countries

Summary Statistics

	Finance-Activity	Finance-Size	Finance-Aggregate	Finance-Dummy	Structure-Activity	Structure-Size	Structure-Aggregate	Structure-Dummy
Mean	4.41	4.07	0 00	0.40	-2.84	-0 99	0 00	0.50
Median	4 85	4 26	0 13	0.00	-2.55	-1.00	0.01	0.50
Standard Deviation	2 62	0 77	1 00	0 50	1.67	0 89	1 00	0.51
Maximum	8 80	5 38	1.73	1.00	-0 76	0 88	1.46	1.00
Minmum	-1 45	2.70	-2.06	0 00	-6 73	-2.80	-2 30	0.00
Observations	42	42	42	42	42	42	42	42

Correlations

	Finance-Activity	Finance-Size	Finance-Aggregate	Finance-Dummy	Structure-Activity	Structure-Size	Structure-Aggregate	Structure-Dummy
Finance-Activity	1							
Finance-Size	0.90	1						
	(0.001)							
Finance-Aggregate	0.97	0.97	1					
	(0.001)	(0.001)						
Finance-Dummy	0.71	0.70	0.72	1				
	(0.001)	(0.001)	(0.001)					
Structure-Activity	0.89	0.65	0.79	0 50	1			
	(0.001)	(0.001)	(0 001)	(0 001)				
Structure-Size	0 52	0 48	0.51	0.14	0 67	1		
	(0 001)	(0.001)	(0.001)	(0.386)	(0 001)			
Structure-Aggregate	0 77	0.61	0 71	0 35	0.91 `	0 91	1	
	(0.001)	(0.001)	(0 001)	(0.024)	(0.001)	(0 001)		
Structure-Dummy	0 60	0.45	0.54	0.34	0.76	0 79	0.85	1
Ť	(0 001)	(0 003)	(0.001)	(0.001)	(0.001)	(0 001)	(0 001)	

p-values are reported in parentheses

Finance-Activity = log(Total value traded as share of GDP * Claims on private sector by financial institutions as share of GDP)

Finance-Size = log(Market capitalization and claims on private sector by financial institutions as share of GDP)

Finance-Aggregate = First principal component of Finance-Activity and Finance-Size

Finance-Dummy = Dummy vanable that takes the value 0 if total value traded as share of GDP and claims on private sector by financial intermedianes as share of GDP are less than the respective

Structure-Activity = log(Total value traded divided by claims on private sector by commercials banks)

Structure-Size = log(Market capitalization divided by claims on private sector by commercials bank)

Structure-Aggregate = First principal components of Structure-Activity and Structure-Size

Structure-Dummy = Dummy variable that takes the value 1 if Structure-Aggregate is above the median, 0 otherwise

Table 3: Industry Growth Across Countries

Summary Statistics

	Industry's real growth		
		number of firms	average size of firms
Mean	3.34	1.41	2.16
Median	2.89	0.83	2.57
Standard Deviation	9.84	8.02	9.88
Maximum	100.00	94.37	41.03
Minimum	-44.74	-41.42	-93.06
Observations	1258	1111	1051

Correlations

	Industry's real growth	Industry's growth in number of firms	Industry's growth in average size of firms
Industry's real growth	1		
Industry's growth in number of firms	0.38 (0.001)	1	
Industry's growth in average size of firms	0.71 <i>(0.001)</i>	-0.42 (0.001)	1

Industry's real growth = annual compounded growth rate in real value added for 1980-90.

Industry's growth in number of firms = log-difference between number of establishments in 1990 and 1980

Industry's growth in average size of firms = log-difference between industry's value added divided by number of establishments in 1990 and 1980

Table 4: Financial Development and Industry Growth

OLS Regressions

	Finance-Activity	Finance-Size	Finance-Aggregate	Finance-Dummy
Interaction (external dependence x Finance-Activity)	1.553 (0.001)			
Interaction (external dependence x Finance-Size)		4.431 (0.001)		
Interaction (external dependence x Finance-Aggregate)			3.866 (0.001)	
Interaction (external dependence x Finance-Dummy)				4.499 (0.001)
R ²	0.281	0.275	0.279	0.266
Number of observations	1222	1222	1222	1222
Differential in real growth rate	2.265	2.169	2.506	1.656

TSLS Regressions

	Finance-Activity	Finance-Size	Finance-Aggregate
Interaction (external dependence x Finance-Activity)	0.790 (0.048)		
Interaction (external dependence x Finance-Size)		2.816 (0.057)	
Interaction (external dependence x Finance-Aggregate)			2.075 (0.052)
Number of observations	1222	1222	1222
Differential in real growth rate	1.152	1.378	1.345

The dependent variable is the annual compounded growth rate in real value added for 1980-90 for each industry in each country. The p-values for heteroskedasticity robust standard errors are reported in parentheses. All regressions also include the industry's share of total value added in manufacturing in 1980 and country and industry dummies. The differential in real growth rate indicates how much faster an industry at the 75th percentile of external dependence grows with respect to an industry at the 25th percentile level in a country at the 75th percentile of the respective measure of financial development compared to a country at the 25th percentile. We use the British, French and German legal origin dummies as instruments for financial development in the TSLS regressions.

Finance-Activity = log(Total value traded as share of GDP * Claims on private sector by financial institutions as share of GDP)
Finance-Size = log(Market capitalization and claims on private sector by financial institutions as share of GDP)
Finance-Aggregate = First principal component of Finance-Activity and Finance-Size

Finance-Dummy = Dummy variable that takes the value 0 if total value traded as share of GDP and claims on private sector by financial intermediaries as share of GDP are less than the respective sample mean, 1 otherwise

Table 5: Financial Structure and Industry Growth

OLS Regressions

	Structure-Activity	Structure-Size	Structure-Aggregate	Structure-Dummy
Interaction (external dependence x Structure-Activity) Interaction (external dependence x Structure-Size)	2.251 (0.001)	2.618 (0.008)		
Interaction (external dependence x Structure-Aggregate)		(5.555)	3.365 (0.001)	
Interaction (external dependence x Structure-Dummy)				4.842 (0.001)
R^2	0.278	0.266	0.274	0.267
Number of observations	1222	1222	1222	1222

TSLS Regressions

	Structure-Activity	Structure-Size	Structure-Aggregate
Interaction (external dependence x Structure-Activity)	0.987 (0.284)		,
Interaction (external dependence x Structure-Size)		-2.167 (0.325)	
Interaction (external dependence x Structure-Aggregate)			-0.380 <i>(0.855)</i>
Number of observations	1222	1222	1222

The dependent variable is the annual compounded growth rate in real value added for 1980-90 for each industry in each country. The p-values for heteroskedasticity robust standard errors are reported in parentheses. All regressions also include the industry's share of total value added in manufacturing in 1980 and country and industry dummies. We use the British, French and German legal origin dummies as instruments for financial structure in the TSLS regressions.

Structure-Activity = log(Total value traded divided by claims on private sector by commercials banks)

Structure-Size = log(Market capitalization divided by claims on private sector by commercials bank)

Structure-Aggregate = First principal components of Structure-Activity and Structure-Size

Structure-Dummy = Dummy variable that takes the value 1 if Structure-Aggregate is above the median, 0 otherwise

Table 6: Financial Development, Financial Structure, and Industry Growth

	Structure-Activity	Structure-Size	Structure-Aggregate
Interaction (external dependence x Structure-Activity)	-4.599 (0.105)		
Interaction (external dependence x Structure-Size)		-1.376 (0.210)	
Interaction (external dependence x Structure-Aggregate)			-2.113 (0.167)
Interaction (external dependence x Finance-Activity)	3.375 (0.008)	1.114 (0.032)	1.476 (0 005)
Number of observations	1222	1222	1222
	Structure-Activity	Structure-Size	Structure-Aggregate
Interaction (external dependence x Structure-Activity)	-4.437 (0.163)		
Interaction (external dependence x Structure-Size)		-1 758 (0.129)	
Interaction (external dependence x Structure-Aggregate)			-2.792 (0.113)
Interaction (external dependence x Finance-Size)	11.842 (0.021)	4.438 (0 015)	6.172 (0.001)
Number of observations	1222	1222	1222
	Structure-Activity	Structure-Size	Structure-Aggregate
Interaction (external dependence x Structure-Activity)	-4.841 (0.118)		
Interaction (external dependence x Structure-Size)		-1.562 (0.163)	
Interaction (external dependence x Structure-Aggregate)			-2.460 (0.131)
Interaction (external dependence	9.254	3.090	4.213

The dependent variable is the annual compounded growth rate in real value added for 1980-90 for each industry in each country. The p-values for heteroskedasticity robust standard errors are reported in parentheses. All regressions also include the industry's share of total value added in manufacturing in 1980 and country and industry dummies. All regressions are TSLS. We use the British, French and German legal origin dummies and the share of Catholic, Muslim and Protestant population in total population as instruments for financial development and financial structure.

Finance-Activity = log(Total value traded as share of GDP * Claims on private sector by financial institutions as share of GDP)
Finance-Size = log(Market capitalization and claims on private sector by financial institutions as share of GDP)
Finance-Aggregate = First principal component of Finance-Activity and Finance-Size

(0.012)

1222

(0.022)

1222

(0 002)

1222

Structure-Activity = log(Total value traded divided by claims on private sector by commercials banks)
Structure-Size = log(Market capitalization divided by claims on private sector by commercials bank)
Structure-Aggregate = First principal components of Structure-Activity and Structure-Size

x Finance-Aggregate)

Number of observations

Table 7: Financial Structure, the Legal Environment, and Industry Growth

	Structure-Activity	Structure-Size	Structure-Aggregate
Interaction (external dependence x Structure-Activity)	-0.153 (0.866)		
Interaction (external dependence x Structure-Size)		0.356 (0.803)	
Interaction (external dependence x Structure-Aggregate)			0.051 <i>(0.974)</i>
Interaction (external dependence x Creditor)	0.177 (0.749)	0.205 (0.715)	0.185 <i>(0.736)</i>
Interaction (external dependence x Anti-Director)	0.014 (0.986)	-0.300 (0.801)	-0.100 (0.928)
Interaction (external dependence x Rule of Law)	0.959 (0.001)	0.875 (0.001)	0.900 (0.001)
F-test Creditor, Anti-Director and Rule of Law	4.37 (0.005)	5.70 (0.001)	5.66 (0.001)
Number of observations	1104	1104	1104

The dependent variable is the annual compounded growth rate in real value added for 1980-90 for each industry in each country. The p-values for heteroskedasticity robust standard errors are reported in parentheses. All regressions also include the industry's share of total value added in manufacturing in 1980 and country and industry dummies. All regressions are TSLS. We use the British, French and German legal origin dummies and the share of Catholic, Muslim and Protestant population in total population as instruments for financial structure and the legal determinants.

Structure-Activity = log(Total value traded divided by claims on private sector by commercials banks)
Structure-Size = log(Market capitalization divided by claims on private sector by commercials bank)
Structure-Aggregate = First principal components of Structure-Activity and Structure-Size

Creditor = index of secured creditor rights

Anti-director = index of minority shareholder rights

Rule of Law = Measure of the law and order tradition of a country.

Table 8: Financial Development, Financial Structure, and the Growth in Number of Firms

	Structure-Activity	Structure-Size	Structure-Aggregate
Interaction (external dependence x Structure-Activity)	-0.765 (0.604)		
Interaction (external dependence x Structure-Size)		0.260 (0.760)	
Interaction (external dependence x Structure-Aggregate)	<u>}</u>		0.140 (0.897)
Interaction (external dependence x Finance-Activity)	1.448 (0.084)	0.982 (0.001)	0.987 (0.013)
Number of observations	1082	1082	1082

	Structure-Activity	Structure-Size	Structure-Aggregate
Interaction (external dependence x Structure-Activity)	-0.907 (0.505)		
Interaction (external dependence x Structure-Size)		0.107 (0.903)	
Interaction (external dependence x Structure-Aggregate)			-0.094 (0.932)
Interaction (external dependence x Finance-Size)	5.631 (0.041)	3.775 (0.001)	3.945 (0.007)
Number of observations	1082	1082	1082

	Structure-Activity	Structure-Size	Structure-Aggregate
Interaction (external dependence x Structure-Activity)	-0.929 (0.513)		
Interaction (external dependence x Structure-Size)		0.181 (0.833)	
Interaction (external dependence x Structure-Aggregate)			0.011 (0.992)
Interaction (external dependence x Finance-Aggregate)	4.105 (0.052)	2.676 (0.001)	2.753 (0.009)
Number of observations	1082	1082	1082

The dependent variable is the log difference between the number of establishments in 1990 and 1980 for each industry in each country. The p-values for heteroskedasticity robust standard errors are reported in parentheses. All regressions also include the industry's share of total value added in manufacturing in 1980 and country and industry dummies. All regressions are TSLS. We use the British, French and German legal origin dummies and the share of Catholic, Muslim and Protestant population in total population as instruments for financial development and financial structure

Finance-Activity = log(Total value traded as share of GDP * Claims on private sector by financial institutions as share of GDP)
Finance-Size = log(Market capitalization and claims on private sector by financial institutions as share of GDP)
Finance-Aggregate = First principal component of Finance-Activity and Finance-Size

Structure-Activity = log(Total value traded divided by claims on private sector by commercials banks)
Structure-Size = log(Market capitalization divided by claims on private sector by commercials bank)
Structure-Aggregate = First principal components of Structure-Activity and Structure-Size

Table 9: Financial Development, Financial Structure, and the Growth in Average Size of Firms

	Structure-Activity	Structure-Size	Structure-Aggregate
Interaction (external dependence x Structure-Activity)	-2.514 (0 191)		
Interaction (external dependence x Structure-Size)		-1 638 (0.067)	
Interaction (external dependence x Structure-Aggregate)			-2.106 (0.075)
Interaction (external dependence x Finance-Activity)	1.714 (0.066)	0.636 (0.068)	0.949 (0.016)
Number of observations	1051	1051	1051

	Structure-Activity	Structure-Size	Structure-Aggregate
Interaction (external dependence x Structure-Activity)	-1.556 (0.399)		-
Interaction (external dependence x Structure-Size)		-1.682 (0 064)	
Interaction (external dependence x Structure-Aggregate)			-2.063 (0.096)
Interaction (external dependence x Finance-Size)	4.257 (0.182)	2.329 (0.058)	3.366 (0.017)
Number of observations	1051	1051	1051

	Structure-Activity	Structure-Size	Structure-Aggregate
Interaction (external dependence x Structure-Activity)	-2.048 (0.285)		
Interaction (external dependence x Structure-Size)		-1.663 (0.065)	
Interaction (external dependence x Structure-Aggregate)			-2.104 (0.081)
Interaction (external dependence x Finance-Aggregate)	3.833 (0.115)	1.693 (0.062)	2.499 (0.016)
Number of observations	1051	1051	1051

The dependent variable is the log difference between the average size of establishments in 1990 and 1980 for each industry in each country. The p-values for heteroskedasticity robust standard errors are reported in parentheses. All regressions also include the industry's share of total value added in manufacturing in 1980 and country and industry dummies. All regressions are TSLS. We use the British, French and German legal origin dummies and the share of Catholic, Muslim and Protestant population in total population as instruments for financial development and financial structure.

Finance-Activity = log(Total value traded as share of GDP * Claims on private sector by financial institutions as share of GDP)
Finance-Size = log(Market capitalization and claims on private sector by financial institutions as share of GDP)
Finance-Aggregate = First principal component of Finance-Activity and Finance-Size

Structure-Activity = log(Total value traded divided by claims on private sector by commercials banks)
Structure-Size = log(Market capitalization divided by claims on private sector by commercials bank)
Structure-Aggregate = First principal components of Structure-Activity and Structure-Size

Table 10: Financial Structure, the Legal Environment, and the Sources of Industry Growth

Growth of the Number of Firms

	Structure-Activity	Structure-Size	Structure-Aggregate
Interaction (external dependence x Structure-Activity)	0 547 (0 489)		
Interaction (external dependence x Structure-Size)		0.505 (0 661)	
Interaction (external dependence x Structure-Aggregate)			0.888 (0 498)
Interaction (external dependence x Creditor)	0 750 (0 136)	0.750 (0 137)	0 755 (0 133)
Interaction (external dependence x Anti-Director)	-0 268 (0 726)	-0 279 (0 760)	-0 467 (0 630)
Interaction (external dependence x Rule of Law)	0 429 (0 117)	0 552 (0 003)	0.452 (0 059)
F-test Creditor, Anti-Director and Rule of Law	1 81 <i>(0 144)</i>	4.04 (0 007)	2 83 (0 038)
Number of observations	997	997	997

Growth of the Average Size of Firms

	Structure-Activity	Structure-Size	Structure-Aggregate
Interaction (external dependence x Structure-Activity)	-0 164 (0 854)		
Interaction (external dependence x Structure-Size)		-0.584 (0 651)	
Interaction (external dependence x Structure-Aggregate)			-0 520 (0 725)
Interaction (external dependence x Creditor)	-0 575 (0 275)	-0 584 (0 271)	-0 581 (0 270)
Interaction (external dependence x Anti-Director)	-0 532 (0.471)	-0 278 (0 787)	-0.338 (0 728)
Interaction (external dependence x Rule of Law)	0 372 (0.253)	0 377 (0 106)	0 406 (0 164)
F-test Creditor, Anti-Director and Rule of Law	2 06 (0.104)	1.84 (0.139)	2 16 (0 091)
Number of observations	970	970	970

The dependent variable is the log difference in the number of establishment (average size of establishments) between 1990 and 1980 in the top panel (bottom panel) for each industry in each country. The p-values for heteroskedasticity robust standard errors are reported in parentheses. All regressions also include the industry's share of total value added in manufacturing in 1980 and country and industry dummies. All regressions are TSLS. We use the British, French and German legal origin dummies and the share of Catholic, Muslim and Protestant population in total population as instruments for financial structure and the legal determinants.

Structure-Activity = log(Total value traded divided by claims on private sector by commercials banks)
Structure-Size = log(Market capitalization divided by claims on private sector by commercials bank)
Structure-Aggregate = First principal components of Structure-Activity and Structure-Size

Creditor = index of secured creditor rights

Anti-director = index of minority shareholder rights

Rule of Law = Measure of the law and order tradition of a country

Table A1: Financial Development, Financial Structure and the Legal Environment Across Countries

Country	Finance-Activity	Finance-Size	Finance-Aggregate	Finance-Dummy	Structure-Activity	Structure-Size	Structure-Aggregate	Structure-Dummy	Creditor	Anti-director	Rule of Law
Australia	6 76						1 18	1	•	4 2	
Austria	5.23							C		2	: 10
Bangladesh	-1 45									_	
Belgium	4 34						0 63	1	_		
Brazil	4 72							1	•		
Canada	6 77				. •••						
Chile	4 23			C				C	_		
Colombia	1 95			C			-0.63	Q		3	2 08
Costa Rica	-0 91									_	
Denmark	4 70	4 09	0 07	C			0 07	1	•		
Egypt	1 70			C				C			
Finland	4 99			1							
France	6 01						-0 45	C			
Germany	7 26	4.67	0 95	1							
Greece	2 59	3 92	-0 46	C	-4 47	-1 62					
India	4 48	3 51			-2 04			C			
Israel	6.37	4 30	0.53	1	-1 32	-0 56	0 76	1	•		
Italy	5 01	4 09	0 13	C	-2 79	-1.57	-0 34	C) 2	1	
Japan	8 80	5.38	1 73	1	-0 77	-0 35		1	2	4	
Jordan	5 85	4 66	0 67	C	-2 21	-0.14	0.73	1		1	
Korea	6 90	4 40	0 70	1	-1 04	-1.03	0.57	1	3		
Malaysıa	6 55	4 87	0 95	1	-1 68	0 11	1 05	1	4	4	
Mexico	3 50	2 81	-1 02	C	-1 27	-0 81	0 62	1	0	1	5.35
Morocco	0 65	3 13	-1 36	Ċ	-5 19	-2.15	-1 49	C			
Netherlands	7 31	4 99	1 18	1	-1 65	-0 75	0 54	1	2		
New Zealand	5 34	4.37	0 38	C	-1 39	0 62	1 46	1	3	4	
Nigeria	-1 12	3 05	-1 76	C	-6 68	-1.41	-1 52	c) 4	3	
Norway	5 75	4 56	0 59	C	-2.44	-1 38	-0 11	C	2		
Pakistan	2 59	3 34	-0 84	C	-3.75	-1 70	-0 73	C) 4	5	
Peru	1 40	2 70	-1.50	C	-2 84	-0.46	0.32	1	0		
Philippines	4 09	3 64	-0,35	C	-2.49	-1.17	0 00	C	0	3	2 73
Portugal	4 23	4 31	0.12	1	-4 26	-2 66	-1 49	C	1	3	
Singapore	7 82	5 35	1 51	1	-1 10	0 39	1 42	1	4	4	8.57
South Africa	6 03	5 23	1 08	C	-2 09	0 88	1 39	1	3	5	4 42
Spain	5 71	4 43	0 49	1	-2 71	-1 55	-0 30	C	2	4	78
Sri Lanka	074	3 23	-1 28	C	-5 09	-0 97	-0 73	C	3	3	6 25
Sweden	6 68	4 83	0.94	1	-1 60	-0 30	0.83	1	2	3	3 10
Turkey	0 98	2 83	-1 50	C	-4 40	-2 10	-1 19	C) 2	2	. 5
UK	7 14				-0 76	0 15	1 38	1	4	5	8 57
US	8 11							. 1	1	5	10
Venezuela	2 51			Ċ			-0 98	c)	1	6 37
Zimbabwe	2.86						0 40	1	4	3	3 68

Finance-Activity = log(Total value traded as share of GDP * Claims on private sector by financial institutions as share of GDP)

Finance-Size = log(Market capitalization and claims on private sector by financial institutions as share of GDP)

Finance-Aggregate = First principal component of Finance-Activity and Finance-Size

Finance-Dummy = Dummy variable that takes the value 0 if total value traded as share of GDP and claims on private sector by financial infermediaties as share of GDP are less than the respective sample mean, 0 otherwise

Structure-Activity = log(Total value traded divided by claims on private sector by commercials banks)

Structure-Size = log(Market capitalization divided by claims on private sector by commercials bank)

Structure-Aggregate = First principal components of Structure-Activity and Structure-Size

Structure-Dummy = Dummy variable that takes the value 1 if Structure-Aggregate is above the median, 0 otherwise

Creditor = index of secured creditor rights

Anti-director = index of minority shareholder rights

Rule of Law = Measure of the law and order tradition of a country

Table A2: Country Classification of Financial Structure

Country	Structure-Activity	Country	Structure-Size	Country	Structure-Aggregate	Country	Structure-Dummy
UK	-0.76	South Africa	0.88	New Zealand	1.46	Australia	1
Japan	-0.77	New Zealand	0.62	Singapore	1.42	Belgium	1
US	-0.86	Singapore	0.39	South Africa	1.39	Brazil	1
Brazil	-0.98	UK	0.15	UK	1.38	Canada	1
Korea	-1.04 ·	Malaysia	0.11	Australia	1.18	Denmark	1
Singapore	-1.10	Australia	0.05	US	1.10	Germany	1
Australia	-1.19	Canada	-0.06	Japan	1.07	Israel	1
Mexico	-1.27	Jordan	-0.14	Canada	1.06	Japan	1
Israel	-1.32	US	-0.24	Malaysia	1.05	Jordan	1
Canada	-1.35	Belgium	-0.27	Brazil	1.03	Korea	1
New Zealand	-1.39	Sweden	-0.30	Sweden	0.83	Malaysia	1
Sweden	-1.60	Brazil	-0.30	Israel	0.76	Mexico	1
Germany	-1.64	Japan	-0.35	Jordan	0.73	Netherlands	1
Netherlands	-1.65	Peru	-0.46	Belgium	0.63	New Zealand	1
Malaysia	-1.68	Zimbabwe	-0.47	Mexico	0.62	Peru	1
India	-2.04	Israel	-0.56	Korea	0.57	Singapore	1
South Africa	-2.09	Netherlands	-0.75	Netherlands	0.54	South Africa	1
Jordan	-2.21	Chile	-0.75	Zimbabwe	0.40	Sweden	1
Belgium	-2.27	Mexico	-0.81	Peru	0.32	UK	1
Norway	-2.44	Denmark	-0.90	Denmark	0.07	US	1
Philippines	-2.49	Sri Lanka	-0.97	Germany	0.02	Zimbabwe	1
Zimbabwe	-2.60	Korea	-1.03	Philippines	0.00	Austria	0
Spain	-2.71	Philippines	-1.17	Chile	-0.06	Banglades	0
Italy	-2.79	Finland	-1.33	India	-0.07	Chile	0
Denmark	-2.80	Costa Rica	-1.34	Norway	-0.11	Colombia	0
France	-2.83	Norway	-1.38	Finland	-0.30	Costa Rica	0
Peru	-2.84	Nigeria	-1.41	Spain	-0.30	Egypt	0
Finland	-3.10	Colombia	-1.47	Italy	-0.34	Finland	0
Chile	-3.46	India	-1.53	France	-0. 4 5	France	0
Austria	-3.55	Spain	-1.55	Colombia	-0.63	Greece	0
Pakistan	-3.75	Italy	-1.57	Sri Lanka	-0.73	India	0
Colombia	-3.86	Germany	-1.59	Pakistan	-0.73	Italy	0
Portugal	-4.26	Venezuela	-1.61	Greece	-0.92	Morocco	0
Turkey	-4.40	Greece	-1.62	Venezuela	-0.98	Nigeria	0
Greece	-4.47	Pakistan	-1.70	Egypt	-1.18	Norway	0
Venezuela	-4.65	France	-1.73	Turkey	-1.19	Pakistan	0
Egypt	-4.82	Egypt	-1.85	Austria	-1.35	Philippines	0
Sri Lanka	-5.09	Turkey	-2.10	Costa Rica	-1.46	Portugal	0
Morocco	- 5.19	Morocco	-2.15	Morocco	-1.49	Spain	0
Costa Rica	-6.65	Portugal	-2.66	Portugal	-1.49	Sri Lanka	0
Nigeria	-6.68	Banglades	-2.66	Nigeria	-1.52	Turkey	Ō
Banglades	-6.73	Austria	-2.80	Banglades	-2.30	Venezuela	0

Policy Research Working Paper Series

	Title	Author	Date	Contact for paper
WPS2365	Leading Indicator Project: Lithuania	Stephen S. Everhart Robert Duval-Hernandez	June 2000	M. Geller 85155
WPS2366	Fiscal Constraints, Collection Costs, and Trade Policies	Keiko Kubota	June 2000	L. Tabada 36896
WPS2367	Gender, Poverty, and Nonfarm Employment in Ghana and Uganda	Constance Newman Sudharshan Canagarajan	June 2000	M. Clarke 31752
WPS2368	Seeds of Corruption: Do Market Institutions Matter?	Harry G. Broadman Francesca Racanatini	June 2000	S. Craig 33160
WPS2369	How the Proposed Basel Guidelines On Rating-Agency Assessments Would Affect Developing Countries	Giovanni Ferri Li-Gang Liu Giovanni Majnoni	June 2000	E. Mekhova 85984
WPS2370	A New Model for Market-Based Regulation of Subnational Borrowing: The Mexican Approach	Marcelo Giugale Adam Korobow Steven Webb	June 2000	M. Geller 85155
WPS2371	Shock Persistence and the Choice of Foreign Exchange Regime: An Empirical Note from Mexico	Marcelo Giugale Adam Korobow	June 2000	M. Geller 85155
WPS2372	Financial Openness, Democracy, and Redistributive Policy	Mansoor Dailami	June 2000	W. Nedrow 31585
WPS2373	Reciprocity across Modes of Supply in the World Trade Organization: A Negotiating Formula	Aaditya Mattoo Marcelo Olarreaga	June 2000	L. Tabada 36896
WPS2374	Should Credit Be Given for Autonomous Liberalization in Multilateral Trade Negotiations?	Aaditya Mattoo Marcelo Olarreaga	June 2000	L. Tabada 36896
WPS2375	Asset Distribution, Inequality, and Growth	Klaus Deininger Pedro Olinto	June 2000	M. Fernandez 33766
WPS2376	The Effect of Early Childhood Development Programs on Women's Labor Force Participation and Older Children's Schooling in Kenya	Michael M. Lokshin Elena Glinskaya Marito Garcia	June 2000	P. Sader 33902
WPS2377	Reforming the Water Supply in Abidjan, Côte d'Ivoire: Mild Reform in a Turbulent Environment	Claude Ménard George Clarke	June 2000	H. Sladovich 37698

Policy Research Working Paper Series

	Title	Author	Date	Contact for paper
WPS2378	Disintegration and Trade Flows: Evidence from the Former Soviet Union	Simeon Djankov Caroline Freund	June 2000	R. Vo 33722
WPS2379	India and the Multilateral Trading System after Seattle: Toward a Proactive Role	Aaditya Mattoo Arvind Subramanian	June 2000	L. Tabada 36896
WPS2380	Trade Policies for Electronic Commerce	Aaditya Mattoo Ludger Schuknecht	June 2000	L. Tabada 36896
WPS2381	Savings and the Terms of Trade under Borrowing Constraints	Pierre-Richard Agénor Joshua Aizenman	June 2000	T. Loftus 36317
WPS2382	Impediments to the Development and Efficiency of Financial Intermediation in Brazil	Thorsten Beck	June 2000	E. Mekhova 85984